Artificial Intelligence in Neurodegenerative Diseases: A Review of Available Tools with a Focus on Machine Learning Techniques

Alexandra-Maria Tăuțan^a, Bogdan Ionescu^a, Emiliano Santarnecchi^b

^a University "Politehnica" of Bucharest, Splaiul Independenței 313, 060042 Bucharest, Romania

^bBerenson-Allen Center for Noninvasive Brain Stimulation, Harvard Medical School, 330 Brookline Avenue, Boston, United States of America

Abstract

Neurodegenerative diseases have shown an increasing incidence in the older population in recent years. A significant amount of research has been conducted to characterize these diseases. Computational methods, and particularly machine learning techniques, are now very useful tools in helping and improving the diagnosis as well as the disease monitoring process. In this paper, we provide an in-depth review on existing computational approaches used in the whole neurodegenerative spectrum, namely for Alzheimer's, Parkinson's, and Huntington's Diseases, Amyotrophic Lateral Sclerosis, and Multiple System Atrophy. We propose a taxonomy of the specific clinical features, and of the existing computational methods. We provide a detailed analysis of the various modalities and decision systems employed for each disease. We identify and present the sleep disorders which are present in various diseases and which represent an important asset for onset detection. We overview the existing data set resources and evaluation metrics. Finally, we identify current remaining open challenges and discuss future perspectives.

Keywords: Neurodegenerative Diseases, Computational Approaches, Machine Learning

Declarations of interest: none.

1. Introduction

Neurodegenerative diseases are a class of neurological disorders where neurons from the central nervous system die or are damaged causing severe disabilities, and eventually death. They are typically encountered in old age. However,

Email addresses: alexandra.tautan@sdettib.pub.ro (Alexandra-Maria Tăuţan), bogdan.ionescu@upb.com (Bogdan Ionescu), esantarn@bidmc.harvard.edu (Emiliano Santarnecchi)

disease onset might appear earlier. In the past years, their incidence increased significantly and it is expected that the increase will continue, as the world's population ages [1]. Neurodegenerative diseases are problematic and can become a burden since their cause is unknown and no cure has been discovered. Treatments are currently targeting the alleviation of symptoms. Due to recent advances in artificial intelligence, a significant help can come from the computational approaches targeting diagnosis and monitoring, e.g., detection of disease onset, characterization of the disease, improvement of the differential diagnosis, quantification of the disease progression, tracking of the medication effects. These tasks can be automated or at least improved with the help of machine learning algorithms.

Scope and target of this work. In this context, the present study proposes an in-depth, large scale, analysis of the existing artificial intelligence capabilities in support of the diagnosis and analysis of the main neurodegenerative diseases. Although a large number of neurodegenerative diseases can be defined [2], we target the ones with the highest prevalence and representative of the neurodegenerative spectrum, namely: Alzheimer's Disease, Parkinson's Disease, Huntington's Disease, Multiple System Atrophy, and Amyotrophic Lateral Sclerosis. To retrieve the existing literature, a total of 46 keywords were used, ranging from "neurodegenerative medical devices", "handwriting parkinson detection"," huntington disease machine learning" to "alzheimer brain imaging machine learning". These included combinations of the disease names, symptoms and analysis methods. The publications were selected based on the relevance attributed by the scholar google search engine, focusing on the most impacting and recent publications. All articles that did not include computational methods or the target keywords were excluded. A summary of the article pool is presented in Figure 1. Overall, we reviewed more than 450 articles. As the graph shows, there is an increasing interest for this topic, which is triggered not only by recent advances in deep learning but also by the promising results achieved so far. Other review works on specific neurodegenerative diseases or specific symptoms are also available. Our study goes beyond prior works by providing a general view of existing capabilities in the field rather than focusing on particular disease cases. For the completeness of our work, the reader is referred to existing reviews of the literature each time a relevant study is available.

Overview of our contributions. The main contributions of this study can be summarized with the following: (i) we provide a global review on existing computational approaches used in the whole neurodegenerative spectrum, (ii) we identify and synthesize a general taxonomy of neurodegenerative disorders, (iii) after analyzing current trends, we propose a taxonomy for computational approaches, (iv) we provide a detailed analysis of the various modalities and decision systems employed for each disease, (v) we identify and present the sleep disorders which are present in various diseases and which represent an important asset for onset detection, (vi) we identify and present the most prominent datasets available for building computational systems together with the evaluation methodologies, and finally (vii) we identify the main capabilities as well as the remaining open questions to be solved by upcoming developments.

Previous resources. Several previous reviews on similar topics have been identified. For Alzheimer's disease, the reader is referred to: Laske et. al [3] for a review on the different methods available for diagnosing AD, Cassani et. al [4] for a review on differentiating stages of AD progression using resting-state-EEG, Bhat et. al [5] presents the recent research performed on automated EEG based diagnosis of AD, Maestu et. al [6] for a review on MEG and EEG biomarkers for AD, Zhang et. al [7] for a review on methods for identifying MCI and AD, the conversion from MCI and the progression of AD, Alberdi et. al [8] for a review on methods for monitoring AD in an unobtrusive way, Pellegrini et. al [9] for a review on machine learning techniques used in neuroimaging for dementia and MCI, Davatzikos et. al [10] for a brief overview of machine learning in neuroimaging. For Parkinson's disease, the reader is referred to: Keijsers et al. [11] for a review on the use of wearable movement sensors for PD detection and severity prediction, van Rooden et al. [12] for a review of identification of PD's subtypes via cluster analysis, Ahlrichs et al. [13] for a review of machine learning approaches for recognizing PD motor symptoms, Stenis et al. [14] for a review of wearable accelerometry-based technology for PD rehabilitation purposes, Pasluosta et al. [15] for a review of existing wearable technologies and the Internet-of-Things concept in support of PD diagnostics and treatment, Kubota et al. [16] for a nontechnical tutorial review of relevant machine learning algorithms for large-scale wearable sensor data in PD, Cummins et al. [17] for a review of speech analysis for health in general that includes also dysarthic PD speech, and Impedovo et Pirlo [18] for a review of dynamic handwriting analysis via pattern recognition for the assessment of neurodegenerative diseases, including PD. In contrast to previous work, apart from presenting the latest methods, we focus on a more general perspective, addressing all modalities and computational approaches in an interconnected way, while investigating the current capabilities of the algorithms. For Huntington's disease, there are currently no review publications with a specific focus on the technical aspects of diagnosis or monitoring. Several publications provide an overview of the methods available for analysis and detection of different gait abnormalities in either neurological or human movement disorders, including HD: Orru et. al [19] provides a review of the different uses of SVM for the identification of imaging biomarkers for neurological diseases in MRI, PET or DTI scans, Figueiredo et. al [20] investigates also the use of SVM, but for identifying gait patterns in human motor disorders, Moon et. al [21] provides a systematic overview of evidence for gait variability in neurodegenerative diseases such as: AD, ALS, HD and PD. For the Amyotrophic Lateral Sclerosis, there is a relative sparsity in computational methods developed and we have identified a single overview providing information on dysarthia in ALS. Tomik et. al [22] analyse both clinical symptoms and the technical methods used for the differential diagnosis based on acoustic features. The current work goes beyond these aspects for a more updated and broader analysis. For Multiple System Atrophy, there are currently no prior literature reviews on the computational techniques used in its monitoring and diagnosis. This holds also for REM sleep behaviour disorder where no overview articles dealing with the computational approaches are available. For restless



Figure 1: Number of considered publications for this review distributed over the publication years.

legs syndrome and periodic limb movement, a systematic review was published by Plante [23] on the use of leg actigraphy for periodic limb movements. In contrast, the current study reviews up-to-date current technology and broader implications on the study of neurodegeneration.

Abbreviations. Throughout the entire paper we will use the following abbreviations (by alphabetic order): AD — Alzheimer's Disease, ANN — Artificial Neural Network, ALS — Amyotrophic Lateral Sclerosis, AUC — Area Under the Curve B — Bradykinesia, CNN — Convolutional Neural Networks, CNS — Central Nervous System, CV — Cross-Validation D — Dyskinesia, DBS — Deep Brain Stimulation, DCNN — Deep Convolutional Neural Networks, DLB — Dementia with Lewy Bodies, DNN — Dynamic Neural Networks, DTI Diffusion Tensor Imaging, ECG — Electrocardiogram, EEG — Electroencephalogram, EMG — Electromyography, EOG — Electrooculogram, ERP — Event Related Potentials, FoG — Freezing of Gait, H&Y — Hoehn and Yahr Scale, HC — Healthy Controls, HD — Huntington's Disease, HMM — Hidden Markov Models, ICA — Independent Component Analysis, kNN — k-Nearest Neighbours, LASSO — least absolute shrinkage and selection operator, LDA Linear Discriminant Analysis, LOO — Leave One Out, LSTM — Long Short Term Memory, MCI — Mild Cognitive Impairment, MRI — Magnetic Resonance Imaging, MSA — Multiple System Atrophy, PCA — Principal Component Analysis, PD — Parkinson's Disease, PET — Positron Emission Tomography, PNN — Probabilistic Neural Network, PSP — Progressive Supranuclear Palsy, PSG — Polysomnography, REM — Rapid Eye Movement, RBD — REM Sleep Behavior Disorder, RF — Random Forest, RLS — Restless Legs Syndrome, ROC - Receiver Operating Characteristic, SPET - Single Photon Emission Tomography, SVM — Support Vector Machine, SVR — Support Vector Regression, T Tremor, UPDRS — Unified Parkinson's Disease Rating Scale.

The remainder of this article is organized as follows. Section 2 presents the relevant definitions and proposes a taxonomy for this review, from both the medical and computational point of view. Further on, each subsequent section deals with the computational approaches encountered for specific neurodegenerative diseases representating a specific category of neurodegeneration. Section 3 looks into Alzheimer's disease. Section 4 reviews Parkinson's Disease. Section 5 analyses Huntington's Disease. Section 6 and 7 deal with Amyotrophic Lateral Sclerosis and Multiple System Atrophy respectively. Section 8 reviews the techniques used for detecting and monitoring sleep disorders encountered in multiple neurodegenerative disorders. Section 9 provides a summary of the datasets and evaluation methods used in testing the majority of the computational approaches presented. Section 10 concludes the review and identifies gaps and future challenges for the field.

2. Definitions and Taxonomy

In this section, we propose a taxonomy for the existing computational approaches for neurodegenerative diseases, both from the medical and computational perspectives. Section 2.1 defines the prominent neurodegenerative disorders along with their symptoms. Section 2.2 discusses the various theories adopted in the literature for the categorization of neurodegeneration and proposes a taxonomy. Section 2.3 defines a taxonomy for the computational approaches highlighting their purpose, the monitored disease and clinical features, along with the data modality used for analysis and diagnosis. Neurodegenerative diseases can be regarded as a class of *neurological disorders* that imply the progressive loss of neurons or subsets of neurons from specific functionalanatomical areas of the CNS [24, 2]. We exclude here the neurological diseases caused by traumas at the level of the CNS. As neurodegeneration can affect many types of neurons and functional areas, their symptomatology is diverse and many different diseases can be defined. Their classification is however controversial as a significant number of symptoms overlap.

2.1. Definitions

Alzheimer's Disease is a progressive age-related neurodegenerative disease characterized by the accumulation of *amyloid plaques* (beta-amyloid protein mixture), *neurofibrillary tangles* (clumps of tau proteins) and a *severe loss of connections* between neurons responsible for memory and learning [25]. Symptoms appear initially as mild memory impairments which can also be confounded with age related memory losses. These progress into severe memory impairments leading up to personality changes, language difficulties, motor difficulties, delusions and hallucinations [26]. Diagnostic criteria include the presence of AD biomarkers assessed through MRI or PET images along with an assessment of dementia symptoms and the degree of cognitive impairment [27].

Dementia with Lewy Bodies is caused by the *accumulation of Lewy bodies* (clusters of alpha-synuclein protein) inside the nuclei of neurons from the cerebral cortex and basal ganglia [25]. Since both neurons involved with memory function and motor control are affected, the clinical symptoms of DLB are very similar to the dementia symptoms of AD and the abnormal movements encountered in PD.

Parkinson's Disease is a motor disorder characterized by the *loss of dopamine producing neurons* through the accumulation of alpha-synuclein proteins. The main clinical characteristics include resting tremor, bradykinesia (a slowing of movements), muscle rigidity, gait and postural disturbances, sleep disorders, tiny handwriting and difficulties when speaking or swallowing [25, 26]. A cure for the disease has not been discovered and current treatments focus on alleviating the symptoms, either through medication, physical therapy or deep brain stimulation. Two severity rating scales are used predominantly in medical practice: *Movement Disorder Society - Unified Parkinson's Disease Rating Scale* (MDS-UPDRS) [28] - rating based on behavior and mood, activities of daily living, motor tasks and therapy effect; *Hoehn and Yahr Scale* [29] - rating based exclusively on gait and posture impairments.

Multiple System Atrophy is a progressive neurodegenerative disease that affects multiple areas of the brain and spinal cord responsible with the coordination of the autonomic nervous system [25, 26]. As DLB and PD, it is also linked to the *accumulation of alpha-synuclein* but in this case in the *glia cells*. Symptoms include bradykinesia, impaired speech, orthostatic hypotension, bladder control problems, abnormal sweating and sleep disorders.

Amyotrophic Lateral Sclerosis is a progressive neurodegenerative disease that *affects motor neurons*. Muscles begin to atrophy as their control is no longer possible. The incipient phases of ALS usually affect the limbs and symptoms rapidly progress to other parts of the body. In the final phase of the disease, the muscles controlling the respiratory system begin to weaken. Death usually occurs within 3 to 5 years from disease onset due to respiratory failure. The most relevant clinical features include: severe motor impairments, muscle twitches, speech impairments, difficulties swallowing [25, 26].

Huntington's Disease is an inherited progressive neurodegenerative disease characterized by a *mutation in the huntingtin gene* that causes motor neurons controlling voluntary movements to die [25, 2]. The symptoms include chorea (uncontrolled movements), abnormal body postures, speech impairments, changes in behavior, emotion, judgment and cognition. Death occurs 10 to 30 years after disease onset. The diagnosis is based on genetic testing and neuroimaging techniques.

2.2. Taxonomy of neurodegenerative diseases

When placing a diagnosis, medical professionals take into account the predominant clinical symptomatology, the topography of the neurodegenerative lesion or a combination of the two. The clinical manifestations are a consequence of the specific neurons and system areas that are affected [2, 30]. For instance, dementia and altered high-order brain functions are linked to the anatomical regions that include the hippocampus, entorhinal cortex, limbic system and neocortical areas. Movement disorders are associated with the damage brought to the basal ganglia, thalamus, brainstem nuclei, cerebellar cortex and nuclei, motor cortical areas and lower motor neurons of the spinal cord. At their incipient manifestation, combinations of these symptoms can be observed in several diseases [30].

It is not yet known what causes or triggers neurodegeneration, while the disease characteristics sometimes overlap and their progression is difficult to predict. In recent years, the traditional method of classifying neurodegenerative diseases based on symptomatology revealed difficulties in the diagnostic process of neurodegeneration, and as a consequence, in finding adequate treatment courses [31, 2]. These difficulties stem from the extent of simultaneous occurrence of both clinincal and neuropathological features defined for separate disorders in one individual at the same time. Armstrong et al. [31] described three models approaching the classification of neurodegenerative diseases: a discrete model, an overlap model and a continuum model. The discrete model implies discrete diseases with little overlap of the clinical and neuropathological features. An overlap model implies a certain degree of overlay in the disease features, while in a continuum model the high degree of overlay of the features can be regarded as a continuous variation of features from one disease to another. Figure 2 presents the overlap of four different clinical feature categories in between the selected diseases: *sleep disorders*, *cognitive* and *behavioral changes*, speech impairments and motor impairments. These can further be broken down into other specific disorders. RBD and RLS appear in diseases that seem to be characterized by alpha-synuclein depositions: DLB, PD and MSA. AD also presents symptoms related to sleep disorders, but in this case they are related to alterations in the sleep/wake cycle of the patient. Cognitive and behavioral changes as an effect of the disease are mostly encountered in dementing disorders such as AD or DLB, but also appear in HD, a disease predominantly characterized by motor dysfunctions. The most relevant cognitive impairments include memory loss and problems with perception. Physicological changes due to neurodegenerative diseases include personality changes (in AD), depression and anxiety (in DLB) and mood changes (in HD). Clinical features related to speech impairments can be regarded as modifications in the lexical content and those related to vocalization. The lexical content of speech is altered in the case of dementing diseases (AD and DLB), while vocalization is different for diseases that are governed by motor dysfunctions. Motor impairments are present in parkinsonism syndromes. Hence bradykinesia, posture and balance dysfunction along with facial muscle rigidity are clinical features of DLB, PD and MSA. HD also presents motor impairments but unlike parkinsonism diseases, it presents hyperkinesia characterized by chorea and tremor.

The nosological approach used by medical professionals to diagnose their patients involves an analysis of the main clinical symptoms along with imaging the lesions, if possible. Computational approaches can be used as an aid in the diagnosis and monitoring of these diseases by tracking different classes of symptoms. They can be used either for disease identification or for monitoring the progression and evaluating different treatment courses, either through medical follow-ups or remote tracking. Hence, *in our taxonomy we approach a classification of neurodegenerative diseases based on the clinical symptomatology*. A symptomatology which can also be tracked for following progression and for disease identification. Following the study of Kovacs et al. [30], we propose three categorizations for neurodegenerative disorders based on their predomi-



Figure 2: Overview of the different clinical features of the most prominent neurodegenerative diseases.

nant clinical characteristics: dementia, abnormal movements and the combination of the two. The disorders characterized by abnormal movements can be further subdivided into hypokinetic and hyperkinetic. In hypokinetic diseases, movements are slowed or diminished, whereas in hyperkinetic disorders, uncontrolled movements appear. The taxonomy is provided in Table 1 (the sources for the prevalence data are the following articles [32, 33, 34] and online documents¹). The list of diseases is not exhaustive, but provides a complete overview from the perspective of existing computational approaches. The prevalence of these disorders is increasing and it is estimated that the number of patients will double by 2050 [1], along with the increase in the older population. Details on the prevalence of each disease is available in Table 1. As neurodegeneration is a process affecting mostly individuals older than 60 years, most data is reported in literature with respect to the elder population. Significant variations are reported between different areas of the globe, with a slightly higher prevalence in low and middle income countries. Out of all neurodegenerative diseases, the dementing ones have the highest prevalence, with AD taking the leading role.

Based on the proposed taxonomy, we have selected several diseases that we considered representative for each disease category. For the dementing disorders, we have chosen to focus on Alzheimer's disease due to its slightly higher preva-

¹https://www.alz.co.uk/research/WorldAlzheimerReport2015.pdf,

diseases, https://emedicine.medscape.com/article/1151013 -overview, and a statement of the statement of

https://www.alzheimer-europe.org/Dementia/Other-forms-of-dementia/Neurodegenerative-inter-forms-of-dementia/Neur

https://www.karger.com/Article/FullText/443738, https://www.valueinhealthjournal.com/article/S1098-3015(18)31696-6/fulltext

lence compared to other diseases in this group (0.6% see Table 1). Although fronto-temporal dementia is the second most encountered dementing disorder, we do not focus on this disorder as most of the symptoms overlap with AD. Some aspects of the differential diagnosis between AD and FTD are covered in Section 3. From the motor hypokinetic disorders, Parkinson's disease has the highest prevalence (0.2% see Table 1) and was included in this survey. Although MSA and ALS (with prevalence of 0.003% and 0.006%, respectively see Table 1) are also hypokinetic disorders, we have decided to include them in this survey due to the paucity of studies using machine learning techniques. As they present similar symptoms to PD, aspects on their differential diagnosis is also included. Huntington's disease was chosen as a representative of the motor hyperkinetic disorders group as it has the highest prevalence compared to similar diseases (0.004% see Table 1).

2.3. Taxonomy of computational approaches

Having analyzed the medical perspectives of neurodegeneration and identified the prominent diseases, we now focus on the existing computational approaches that come in support of the diagnosis, monitoring and improvement of the patient's life. As previously mentioned, we shall focus on a symptomatology*based* analysis. Table 2 illustrates the proposed taxonomy of existing approaches. We propose a classification based on: *clinical symptomatology* and the *disease* they characterize or detect, *basic modality* used as input in the computation and their goal. The symptomatology is divided into five main categories: (1) Sleep disorders — which can be further subdivided into several disorders. REM sleep behavior disorder (RBD) and restless leg syndrome (RLS) (see Section 8). (2) Speech impairments — are observed in both dementing and motor debilitating neurodegenerative diseases, however their manifestation is different. In dementing disorders, the lexical content of the speech is altered. In motor disorders, the muscles controling speech production are affected and thus vocalization impairements are present. (3) Motor impairments — the most visible effect of motor impairments is the effect they have on limb muscle control. Thus problems with gait, tremor and posture are very often encountered. Other symptoms include reduced facial expressions and modifications in handwriting (see Sections 4, 5 and 7). (4) *Biomarkers* — the identification of specific protein depositions in specific anatomical locations by analyzing medical images can define disease biomarkers. Modalities such as EEG, EMG or eye movements can also be used as modalities to extract disease biomarkers.

Computational approaches can also be divided based on the disease they are applied to. Here we chose to look only at representative categories of neurodegeneration as presented in Section 2.2. AD is representative for dementing diseases. Frontotemporal dementia was not considered in this review as it is similar in symptoms to AD, while having a lower incidence rate. PD, MSA, ALS and HD are all predominantly motor disorders. Computational methods are highly dependent on the type of data used as input, especially when looking at feature extraction. The sensor modality used for characterizing a disease depends on the type of symptoms analyzed. Hence sleep disorders are usually

	Class	Disease	Lesion Topography	Prevalence	* Main Clinical Symptoms	#Article
	Dementinø	Alzheimer's Disease	Cerebral cortex, Hip- pocampus, Basal nu- cleus of Meynert	0.6% [2015]	Personality changes, cognitive and memory impairments, delusions, hallucinations	135
	0	Frontotemporal De- mentia	Frontal and temporal lobes of the cerebral cor- tex	0.02% [2013]	Altered personality, apathy, disinhibition, impaired memory, planning, attention, per- ception	NA
	Jementing and abnormal	Lewy Body Dementia	Cerebral cortex, Basal ganglia	0.002% [2016]	Cognitive impairments, delusions, depression, anxiety, rigidity, mask-like face	NA
	movements	Corticobasal Degen- eration	Cerebral cortex, Basal ganglia	0.006% [2013]	Language impairment, muscle twitches, ab- normal posture	NA
		Parkinson's Disease	Basal ganglia	0.2% [2017]	Slowing of voluntary movements, muscle rigidity, resting tremor, difficulty speaking, gait and postural disturbances, tiny hand- writing, sleep disorders	259
s		Olivoponto cerebel- lar atrophy	Cerebellum, Pons, Infe- rior olives	0.005%	Ataxia, tremor, rigidity, sleep disorders, de- pression, tremor	NA
anəməvoM		Progressive Supranu- clear Palsy	Cerebral Nuclei	0.006% [2013]	Loss of balance, difficulty moving eyes, slowing of movement, slurred speech, per- sonality changes	NA
IsmrondA	Hypokinetic	Multiple System At- rophy	Several areas of the brain and spinal cord	0.003% [2013]	Low blood pressure when standing up, ab- normal breathing during sleep, difficulty urinating, abnormal sweating, slowness of movement, impaired speech	9
		Amyotrophic Lateral Sclerosis	Spinal cord	0.006% [2013]	Weakening of the muscles, sleep disorders, involuntary uncontrolled sighing, problems swallowing	15
	Hyperkinetic	Huntington's Disease	Basal ganglia (caudate nucleus, corpus stria- tum)	0.004%	Uncontrolled movements, abnormal body posture, changes in behavior and cognition	50
		Essential Tremor	Basal ganglia	0.003%	Tremor of the hand, head, arms, voice, tongue, legs	NA

Table 1: Taxonomy of nerodegenerative diseases: classification based on symptoms, specific disease examples, affected areas, prevalence (*percent normalized to 100,000 people), clinical symptoms indicative of the overlap between diseases, and prevalence of sources cited in this article.

characterized by PSG and actigraphy based recordings. Speech impairements are analyzed through voice recordings of participants performing different tasks, while motor impairements through a variety of sensors that measure movement in controlled and uncontrolled settings, e.g., accelerometers, EMG, videos. The purposes of using computational methods in helping patients suffering from neurodegenerative disorders is many fold. They can aid in the diagnostic by providing symptom characterization and help in identifying the exact disease. They can also be used for providing an objective monitoring method of disease progression. As the exact classification of neurodegenerative diseases is difficult when symptoms overlap, automatic methods of identifying small symptomatological differences are desired. A promising use of computational methods is the forecasting of events related to disease symptoms or even the identification and classification of the disease prior to the clinical onset. The neurodegenerative process can sometimes start years before symptoms are observed.

3. Alzheimer's Disease

In this section, we present a review of the most widely used computational approaches in the diagnosis and monitoring of Alzheimer's disease (reported metrics are presented in Section 9.4). Most studies included in the review focus on the differentiation between AD and healthy controls (baseline participants, usually age-matched), AD and its prodromal state mild cognitive impairment (MCI) or the differential diagnostic between AD and other forms of dementia. There is also significant interest in monitoring the disease progression by determining several levels of severity or trying to predict the conversion from MCI to AD.

3.1. Biomarkers

3.1.1. Use of brain imaging

Alzheimer's disease is linked to the accumulation of alpha-synuclein in the brain tissue. This accumulation can be tracked and studied through different neuroimaging techniques. Currently, neuroimaging methods are the most accurate option for providing an AD diagnosis while the patient is still alive. The golden standard for a precise AD diagnosis remains the autopsy. Since AD is the most prevalent type of neurodegenerative disease, its characteristics as revealed by neuroimaging have been extensively studied resulting in a high availability of large datasets. Topics addressed in literature include: (i) the detection of AD patients from HC [35, 36, 37], (ii) measuring disease severity [38], (iii) helping with the differential diagnostic from different types of dementia [39, 40] and, the most addressed topic, (iv) differentiating between MCI, AD and HC along with the prediction of conversion from MCI to AD.

Detection of AD patients. Several recent studies have addressed the problem of differentiating between AD and HC by using deep learning techniques previously developed for other image processing problems. Islam et. al [36] compare the deep convolutional neural network Inception V5 model with the GoogleNet on MRI data from the OASIS dataset containing 100 AD patients

Table 2: Taxonomy of computational approaches: clinical features or disease they deal with, diseases sharing these features, used sensor modalities and purpose of the approach (PD — Parkinson's disease, DLB — Dementia with Lewy Bodies, MSA — Multiple System Atrophy, AD — Alzheimer's disease, ALS — Amyotrophic Lateral Sclerosis, HD — Huntington's disease).

Clinica	al feature	Disease (s)	Sensor Modality	Purpose
Sleep disorders	REM sleep be- havior disorder	PD, DLB, MSA	Polysomnography, actigraphy, EEG, EMG	diagnosis
	Restless leg syn- drome and peri- odic limb move- ment	PD, DLB, MSA	Polysomnography, actigraphy, EMG	diagnosis
	Disturbed sleep/wake cycle	AD	Polysomnography, actigraphy	progression monitoring
Speech Impair- ments	Lexical content	AD	Voice	diagnosis, progression monitoring
	Vocalization	AD, DLB, MSA, PD	Voice	diagnosis, classifi- cation, progression monitoring
Motor impair- ments	Gait, Freezing of Gait, Posture	PD, DLB, MSA, ALS, HD	Accelerometers, gyro- scopes, force sensors, EMG, video	diagnosis, classifi- cation, progression monitoring, disease identification
	Tremor	PD, DLB, MSA, ALS, HD	Accelerometers, gyro- scopes, EMG, actigra- phy	diagnosis, classifi- cation, progression monitoring
	Facial expres- sions	PD, DLB, MSA	Video, EMG	diagnosis, disease iden- tification
	Bradykinesia	PD	Accelerometer, gyro- scopes	diagnosis, disease iden- tification
	Handwriting	PD	images of handwrit- ing, writing kinemat- ics, EMG, accelerome- ter	diagnosis, disease iden- tification
Biomarkers	Imaging	AD, DLB, PD, MSA, ALS, HD	MRI, PET, SPECT, DTI	diagnosis, classifi- cation, progression, monitoring
	Other biomark- ers	AD, DLB, PD, MSA, ALS, HD	EEG, Eye movement tracking, EMG, ge- netic information, pro- teomics	diagnosis, classifi- cation, progression, monitoring

and 300 HC. Using a 5-fold cross validation, an accuracy of 73.75% was obtained. Katako et. al [35] work on FDG-PET data from the ADNI database using an SVM classification algorithm in a 10-fold cross-validation. A sensitivity of 84% and a specificity of 95% is obtained for differentiating between AD and HC. Sarraf et. al [37] use LeNet-5 resulting in an accuracy of 96.85% for AD vs. HC differentiation.

Measuring AD severity. The severity of AD can also be classified through neuroimaging data. Mahmood et. al [38] use MRI data from a total of 687 AD patients from the OASIS dataset. These are classified into the following classes: no dementia, very mild AD, mild AD and moderate AD with an overall accuracy of approximately 90%.

Differential diagnostic. Differentiating between AD and Fronto-temporal dementia based only on symptoms can be problematic. Davatzikos et. al [39] use voxel based and high dimensional pattern classification features extracted from grey and white matter regions of brain MRI. By using an SVM classifier with a leave-one-out cross-validation, the proposed algorithm can distinguish between AD and FTD with an accuracy of 84.3%. The difficulty in placing a correct AD diagnosis is valid also for other dementing disorders. The correlation between the golden standard test for AD diagnosis, the autopsy, and the data collected from MRI and neurophysiological tests several years prior to death, has been studied by Kautzky et. al [41]. A classification model was built on the collected data using the labels placed after autopsy. A random forest model was created in a 5-fold cross-validation scenario and resulted in an accuracy of only 62%.

Differentiating between AD and MCI. In some patients, mild cognitive impairment is a prodromal symptom of Alzheimer's disease. The differences in the brains of MCI and AD patients along with the conversion of MCI into AD has been extensively studied using brain imaging technology. Most studies make use of MRI data [42, 43, 44, 45, 46]. Some works study the differences between MCI and AD patients by combining multiple imaging technologies such as MRI and PET [47] or FDG-PET [48] or MRI and DTI [49]. Biomarkers for the conversion are proposed by extracting voxel-based features [42], morphometric and volumetric features [50]. Classification is performed using a variety of wellknown algorithms adapted from other applications.

In the early work of Plant et. al [42], brain changes appearing in MCI as predictors of AD are characterized by voxel based features. These features are used together with an SVM classifier allowing a differentiation of MCI from AD with an accuracy of 97.48%. An SVM classifier is also used by Salvatore et. al [43] to differentiate between MRI images obtained from MCI patients that converted to AD and MCI patients that did not convert. Using a nested cross-validation the accuracy was of 66%. Yan et. al [49] fuses MRI and DTI information for the differentiation between subjective cognitive decline, mild cognitive impairment and Alzheimer's disease. The result of an SVM classification is an accuracy of 98.58% for AD vs. HC, of 97.76% for MCI vs. HC and of 80.24% for subjective cognitive decline vs. MCI.

Deep neural networks have been gaining popularity in the field of imaging

classification. Naturally, some of the methods have been adopted in the problem of AD vs. MCI classification based on brain imaging. Ahmed et. al [46] use a 3D convolutional autoencoder network for AD vs. MCI vs. HC classification based on anatomical features. The training set consisted of 210 patients from the ADNI dataset, while the test set was a selection of 30 patients from the CADDementia dataset. The result was a sensitivity of 100%, 80% and 47% for AD, MCI and HC classes respectively. Jabason et. al [45] have also used deep autoencoders for feature selection. With a 5-fold cross validation used on data from the ADNI dataset the accuracy, sensitivity and specificity obtained was of 98.55%, 98.79% and 99.31% respectively.

3.1.2. Use of EEG

As Alzheimer implies a severe loss of neuronal connections, changes can also be observed on the recorded EEG of AD patients. When compared to healthy controls, EEG signals recorded from AD patients show a slowing down of the characteristic EEG frequency bands and a decrease in complexity due to the diminished neuronal synchronization and of different types of oscillations [51]. Due to the non-invasive nature of the recording, EEG is a good candidate for the extraction of AD biomarkers. EEG based biomarkers have been used in literature to: (i) automatically classify AD patients and HC [51, 52, 53], (ii) to provide help in the differential diagnostic between AD and other types of dementia [54, 55], (iii) to automatically distinguish between Mild-Cognitive Impairment (MCI) and different stages of AD [53, 56].

Classification of AD patients. Most work conducted on the automatic detection of AD vs. HC is focused on extracting computational biomarkers based on the slowing down of EEG frequencies and the reduction in signal complexity. Trambaiolli et. al [52] takes advantage of the slowing down of EEG activity by extracting spectral and coherence features from EEG data and using them as input to an SVM classifier. Feature selection techniques are used to increase the performance on a dataset of 22AD and 12HC with a leave-one-out validation method. The classification accuracy was of 91.8%. Automated EEG based AD classification with a low-density EEG montage has been proposed by Cassani et. al [51]. Using only seven EEG channels, the data was pre-processed using ICA and wavelet decomposition for artifact removal. Several groups of features were extracted: spectral, coherence and amplitude modulation features. These were employed in a 10-fold cross validation framework for an SVM classified. The performance of the model was evaluated using accuracy, sensitivity and specificity (77.3%, 79.2%, 75.2% respectively).

Differential diagnosis. Placing a diagnosis of Alzheimer is not always easy as the symptoms most often overlap with other types of dementia. Several studies have proposed solutions for helping the differential diagnosis. Dauwan et. al [54] use quantitative EEG features combined with clinical and neurophysiological information, visual EEG and cerebrosinal fluid diagnostic information for creating a model that differentiates between DLB and AD. The proposed model uses a random forest classifier and was tested on 66DLB, 66AD and 66HC subjects. An accuracy of 87% was obtained for the differential diagnostic problem. The differentiation between PD related dementias and AD is studied by Jeong et. al [55] using an LDA classifier on features extracted from wavelet energy and coherence. The differentiation between conditions was obtained with an accuracy of 79.18% revealing significant differences in the beta and gamma bands.

Differentiating MCI from AD. The majority of studies extracting AD biomarkers from EEG, use resting-state EEG recordings of different lengths. Mamani et. al [53] proposed an event related potential (ERP) based study using an N-back memory task for obtaining AD and MCI biomarkers. A statistical analysis showed a significant difference between AD, MCI and HC on EEG channels recorded over the fronto-centro-parietal part. The problem of classifying HC, MCI and AD subjects has been addressed by McBride et. al [56] based on Sugihara causality. The three class problem has been solved using an SVM classifier in a leave-one-out scenario using a small database of 15HC, 16MCI and 17AD. The best accuracy was of 95.8%.

3.2. Speech Analysis

The effects of Alzheimer's related dementia can also be observed in the speech of patients. Unlike in the case of Parkinson's, where muscles controlling the production of speech are affected, Alzheimer's Disease affects the content of the speech of AD patients. The majority of studies analyzing the speech of AD patients focus on features related to the semantics of the spoken communication. Topics of interest include: (i) the automatic differentiation between AD and HC [57, 58, 59], and (ii) the detection of the prodromal stage of MCI and its different intermediate stages [60, 61].

Differentiating between AD and HC. A method for the automatic detection of AD subjects using the semantic content of speech was proposed by Fraser et. al [58]. A total of 370 features were extracted from the Dementia-Bank database using as input the syntactic complexity, the grammatical constituents, the psycholinguistics (frequency of certain words), vocabulary richness and repetitiveness but also features derived from the acoustic properties of speech. By using a multi-linear regression in a 10 fold cross-validation scenario a maximum average accuracy of 81.92% was obtained. Konig et. al [57] looked at semantic fluency in 93 AD and MCI patients versus 24 HC using an SVM classifier with a leave-one-out approach. The result was an accuracy of 93.9%. De Ipina et. al [62] aimed at extracting biomarkers of AD from speech, both from spontaneous speech as well as analyzing the emotional response from acoustic features. The best classification accuracy was of a 97.7% using an SVM classifier that had as input emotional features as well.

Detection of prodromal AD from MCI. Automatically identifying patients suffering from mild cognitive impairments can be useful for the prediction of conversion to AD. In this case the semantic features of speech can also be of help. In the work of Konig et. al [60], the distinction between MCI, AD and HC is studied through the extraction of semantic, vocal and statistical features from a short vocal task. Using random sub-sampling for data balancing and an SVM classifier, HC are distinguished from MCI subjects with 79% accuracy. Satt et. al [63] employs data regularization techniques to overcome data sparseness from a small database of 15HC, 23MCI and 26 AD subjects. Semantic, vocal and acoustic features are used with a Naive Bayes classifier. The differentiation between MCI and HC reaches an accuracy of 80% while MCI and AD reaches 87%.

3.3. Eye Movement Analysis

Detecting patients in the mild cognitive impairment state that could convert to AD is useful for early treatment intervention and better disease management. Existing methods make use of the different eye movement patterns resulting as a reaction to different visual stimuli. Pavisics et. al [64] use eye tracking related features, e.g., number of saccades, wave jerks, maximum fixation duration, to distinguish between AD and HC. By employing a Hidden Markov Model an accuracy of 95% is obtained. Eve tracking was also used by Parsons et. al [65] to distinguish between AD and posterior cortical atrophy. Using a Hidden Markov Model to model movements in gaze location, a differentiation accuracy of 95.5%is obtained. Alzheimer's disease leads to severe cognitive impairments and the emotional toll it takes on the patients should not be ignored when proposing different treatment courses. Chung et. al [66] analyzes visual scanning behavior to automatically detect apathy in AD patients. Two separate LSTM cells are used to model visual scanning behavior during emotional and non-emotional stimuli presentation. The output of the recursive neural network is fed into a logistic regression classifier with an outcome of 74% AUROC within a hold out validation.

3.4. Gene Analysis

The development of Alzheimer's disease in some individuals has also been linked to a certain genetic predisposition. The human genome contains a high amount of data unique for each individual. Computational methods, more specifically machine learning tools, have proved to be extremely useful in mapping this information and determining specific genetic links to diseases. AD is no exception and several works focus on identifying genes or gene interactions related to AD development. For instance, Park et. al [67] studied the genetic interactions that could be correlated to AD. The input data was fed into a Random Forest classifier to detect HC and AD related information. An accuracy of 90.2% was obtained. Huang et. al [68] aimed at identifying genes highly correlated to Alzheimer's disease from the whole genome. Genes were labeled as AD and non-AD related. By extracting several genome related features, an SVM classifier with a radial basis kernel was used. The receiver operating characteristic was of 84.56%. Xu et. al [69] also uses protein sequence information with an SVM classifier, resulting in an accuracy of 85.7% in predicting AD.

3.5. Multimodal Features

Alzheimer's disease, along with all the other neurodegenerative diseases, is a complex disorder that affects many facets of the normal functioning of a patient.

Using one type of modality as input for analyzing the disease might in some cases be sufficient but in most cases is not enough for an adequate diagnosis. Therefore, some researchers focused on harvesting information from complementary sources. For instance, Alvarez et. al [70] proposed the ICT4LIFE platform to monitor the behavior of AD. For a more accurate disease classification, information is obtained from multiple sources including electronic health records, body sensors and Kinect sensors. Several features are extracted and a Sparse Autoencoder is used for optimizing feature selection. The result is classified with a logistic regression with an accuracy, precision and recall of 98.4%, 98.7% and 98.3% respectively. Colloby et. al [71] used a combination between EEG and MRI data to distinguish between AD and DLB. Using a SVM classifier, it achieves an accuracy of 90%. The fusion of different technologies has also been investigated in this case. Fraser et. al [72] combined eye movement analysis with speech features in a logistic regression model. Using the two types of features an accuracy of 86% was obtained for classifying MCI and HC subjects. Grassi et. al [73] looks at the conversion of MCI to AD within a 3 year time frame using sociodemographic characteristics, clinical and neurophysiological test scores. With a SVM classifier, an AUC of 0.962 is obtained.

3.6. Summary

Alzheimer's disease symptoms and progression can be investigated and tracked through diverse methods. In this section, we have covered some of the topics that use computational methods in the study of Alzheimer's disease. With specific regard to machine learning techniques, these can be applied on problems with a diverse scope, ranging from AD detection to differential diagnosis and predicting disease progression. A selection of some of the most relevant machine learning research works on AD presented in this review is summarized in Table 3. The information contains details on the purpose of the study, the sensor modality, the type of classifier used and the best performance obtained.

4. Parkinson's Diseases

Parkinson's disease is the second most encountered neurodegenerative disease and we consider it to be representative for hypokinetic diseases with similar symptoms. PD and other hypokinetic diseases are characterized by *bradykinesia*, *muscle rigidity* and *freezing of movements*. As the disease progresses, different impairments related to difficulties in *muscle control* can be seen in patients. Most computational methods focus on, either detecting PD versus healthy control subjects, or on mapping the differently computed features to diseases measurement severity scales, such as UPDRS or H&Y. PD subtype classification is also of interest.

4.1. Motor Symptoms Monitoring

The most predominant symptoms that affect PD patients are the *motor* disabilities. Depending on the symptoms monitored and the final goal of the

Purpose	Modality	Dataset	Classifier	Eval	Acc	Ref
AD vs. HC	MRI	$\begin{array}{c} 100 \mathrm{AD}, \\ 300 \mathrm{HC} \end{array}$	InceptionV5, GoogleNet	5fold CV	73.75%	[36]
AD vs. HC	MRI	28AD, 15HC	LeNet-5	hold out	96.85%	[37]
AD vs. HC	Eye tracking	26AD, 21HC	HMM	CV	95.5%	[65]
AD vs. HC	EEG	22AD, 12HC	SVM	LOO	91.8%	[52]
AD vs. HC MCI vs. HC	MRI, DTI	28AD, 45MCI, 38AD	SVM	CV	98.58% 97.76%	[49]
AD vs. MCI	EEG	17AD, 16MCI	SVM	LOO	95.8%	[56]
AD vs. MCI	Speech	26AD, 23MCI	SVM	LOO	93.9%	[57]
preAD vs. MCI	Speech	26AD, 23MCI	Naive Bayes	-	80%	[63]
MCI vs. HC	Eye tracking, speech	27MCI, 30HC	Logistic Regression	LOO	86%	[72]
AD vs. DLB	EEG	66AD, 66DLB	RF	-	87%	[54]
AD vs. DLB	EEG, MRI	30AD, 21DLB	SVM	-	90%	[71]
AD vs. FTD	MRI	37AD, 12FTD	SVM	LOO	84.3%	[39]

Table 3: Overview of the most relevant research works using machine learning in handling Alzheimer's disease that were presented in this work. Brief details are provided on the dataset size and content, classification techniques and evaluation methods.

research, different recording and processing methodologies are used. As motor symptoms are some of the most encountered problems in PD, an abundance of studies are available on the topic. Computational methods developed for the analysis of PD motor symptoms aim at discriminating between PD and HC [74] but also at objectively quantifying the severity of the disease though comparisons to the UPDRS and H&Y scales [75, 76, 77]. Another application is the monitoring of the on/off medication states of patients [78]. Analyzing the severity of motor disability with respect to the time passed from the last medication intake, one can suggest adjustments to the treatment scheme.

4.1.1. Gait and posture

Most studies looking at the gait and posture of PD patients focus on distinguishing or characterizing different signals collected for patients with respect to HC.

Recording methods. The dynamics of gait are, in the majority of cases, characterized using wearable *accelerometer sensors* placed at different locations on the body, e.g., ankles [79], waist, wrist [80]. In this type of analysis, video recordings annotated by specialists are considered the golden standard [80]. Another approach for monitoring problems with gait is through *force sensors* placed under the foot [76, 74]. This technology is particularly advantageous as sensors can easily be placed inside the shoe with no significant inconvenience to the user. EMG signals can also be used for abnormal gait detection, however these are more cumbersome to record and integrate in wearable technologies. Kugler et al. [81] used EMG sensors on the lower limb muscles for proposing

objective measures of evaluating gait in standard tests. Impaired balance is also studied, for instance Stack et al. [80] use inertial sensors. Protocols for recording involve mostly walking for a specific distance or standard gait tests used by medical professionals.

Classification techniques. Most analysis performed focus on extracting time and frequency domain features for disease state classification. Statistical, entropy and energy features are predominantly extracted from the time domain signals. In the frequency domain, the predominant frequency is characterized along with the phase and the energy content. Asuroglu et al. [76] introduce a locally weighted Random Forest classification for estimating the severity of PD in comparison to the UPDRS scale, using eight force sensors. Alam et al. [74] add swing and stride time along with statistical time domain features as input to an SVM cubic kernel classifier which distinguishes between PD and HC with an accuracy of 93.6%. Three different studies amounting a total of 93 PD and 73 HC subjects with ground reaction force recordings [82] were used by Zhao et al. [83] for implementing a two-channel model combining LSTM and CNN.

4.1.2. Bradykinesia

Studies focusing exclusively on the analysis of bradykinesia in PD patients estimate the severity of the symptoms based on accelerometer and gyroscope sensors placed on different locations of the body [84, 85]. For instance, Manzanera et al. [85] use a Shimmer platform containing accelerometers, gyroscopes and magnetic sensors to record data from 25 PD and 10 HC subjects while performing a series of standardized motor tasks. The obtained signals were fused and features were computed both in time and frequency domain. After applying a t-test based forward selection wrapper for feature reduction, the remaining features were fed into an SVM classifier. The best results were obtained using a combination of seven features and resulted in error rates as low as 9.3-9.8%. Sama et al. [84] use a support vector regression for classifying symptom severity for a smaller database containing 12 PD patients with an accuracy for bradykinesia detection of 90%.

4.1.3. Freezing of gait

Akinesia occurs in some PD patients with a frequency dependent on the severity of the disease. It is possible to provide support to those suffering from a freezing of gait episode to surpass the moment [86]. Hence many studies focus on the detection of such episodes.

For rehabilitation purposes, the detection should provide good performance in uncontrolled environments with as little intrusion as possible. Most of the studies focus on the detection of freezing of gait episodes using *inertial sensor* based wearable technologies. These include accelerometers and gyroscopes [87] placed on the waist [88], thigh [86] or wrist [89]. Gait specific features are extracted from the time domain, e.g., statistical measures [89, 90], step and stride time and length [91], and frequency domain, e.g. freezing index [86], power in different frequency bands [89]. Good results were obtained in several studies using the SVM classifier [88, 91]. Deep learning models are also investigated. Camps et al. [87] used a six layer convolutional network for FoG detection on 21 PD patients. The deep learning framework achieved 90% for the geometric mean between sensitivity and specificity. Since smartphones already incorporate these recording modalities and are ubiquitously available, their performance in this circumstance has also been evaluated by some studies [90]. EEG signals have also been used by Handojoseno et al. [92] for FoG onset detection. Their dynamics were analyzed with wavelet transform based entropy measures and a back propagation neural network classification.

4.1.4. Tremor

A characterization of disease severity or of treatment efficiency can also be provided by the assessment of tremor severity. A convenient method of assessing tremor is through the analysis of signals obtained from accelerometers incorporated in wearable technology. Rigas et al. [93] propose the use of Hidden Markov Models on accelerometer signals obtained from different parts of the body of 18 PD and 5 HC subjects in different resting conditions. Tremor severity was assessed with an accuracy of 87%. Kostikis et al. [94] used a smartphone for training machine learning algorithms to distinguish the severity of parkinsonian tremor on a database of 25 PD and 20 HC participants. A bagged ensemble of decision trees provided the best results with 82% of the patients being classified correctly.

4.1.5. Dyskinesia

A side-effect of levodopa medication used for alleviating PD symptoms is the appearance of uncontrolled movements. The severity of the unwanted effects of medication can also be tracked. Chelar et al. [95] use magnetic motion trackers to quantify the complexity of involuntary movements present in 10 dyskinetic PD, 10 non-diskinetic PD and 10 HC participants with the help of multiscale entropy. Automatic recognition of dyskinetic episodes was performed using multilayer perceptrons. Inertial sensors are a more natural choice for dyskinesia assessment. Tsipouras et al. [96] used accelerometers and gyroscopes placed at the wrists, legs, chest and waist on a similar sample size to automatically recognize dyskinetic patients using an artificial neural network with one hidden layer and time and spectral features as input. The result was an average classification accuracy of 84.3%.

4.1.6. Multiple-symptoms - "on/off" state detection

The approaches presented above tackle the detection and characterization of only one specific PD motor symptom at a time. However, patients most often experience a combination of motor symptoms at a time. Tracking a combination of the symptoms in different environments and with different medication intake can be problematic.

Controlled environment - on/off detection. For an accurate assessment of the patient's state and of the disease progression, as many motor aspects as possible should be considered in more complex systems. In the early work of Patel et al. [77], the Shimmer platform was used with accelerometers placed on

both lower and upper limbs to classify the severity of tremor, bradykinesia and dyskinesia. Standardized motor tasks were performed according to the motor section of the UPDRS. A total of 31 feature combinations were extracted and fed into seven SVM classifiers with different kernels. The lowest mean estimation error was of 1.2%.

Ambulatory setting - on/off detection. Salarian et al. [97] use a miniature gyroscope placed on the upper limbs for estimating tremor and bradykinesia but this time in an ambulatory setting. The algorithm made use of spectrum analysis for tremor detection and the hilbert transform for bradykinesia estimation. The tremor detection showed an overall sensitivity and specificity of 99.5% and 94.2% respectively. Cole et al. [98] tested several dynamical machine learning techniques. Dynamic support vector machines and Hidden Markov Models had both error rates below 10%. In general, focus is placed on the development of unobtrusive systems for monitoring in uncontrolled environments. An important achievement is the REMPARK system that was designed for long term home monitoring of PD patients. It comprises accelerometers and gyroscope sensors placed on the wrist and waist of the patients. Bayes et al. [78] validate REMPARK in detecting on-off states of 41 PD patients with 97% sensitivity and 88% specificity.

Monitoring of Deep Brain Stimulation Effects. Another lead is to estimate the effects of DBS treatments. The work of Zwartjes et al. [75] investigates the daily activities of 6 PD and 7 HC patients and extracts information for quantification of tremor and bradykinesia. After applying a Decision Trees based activity recognition algorithm, features are extracted for characterization of rest and kinetic tremor, bradykinesia and hypokinesia and threshold-based algorithms are applied. Kinetic tremor was identified with an accuracy of 78.7% during sitting and 81.7% during standing. Angeles et al. [99] evaluated rigidity, tremor and bradykinesia with the goal of DBS treatment optimization. 7 PD subjects performed specific motor tasks with 3D accelerometer, gyroscope and magnetometer sensors placed on the most affected hand. Several classification algorithms were used to achieve an average accuracy of 90.9%.

4.2. Speech Monitoring

As PD causes the loss of neurons in the basal ganglia, the muscles involved in the production of speech are also affected by the same symptoms as the other motor muscles, i.e., rigidity, hypokinesia, and tremor. This causes a great majority of PD patients to have dysarthic, abnormal speech [100]. Although the difficulties in speech production can be regarded as another motor symptom, we have decided to describe the computational approaches used on abnormal PD speech separately due to the abundance of literature and the different sensor modalities that are used as input. Dysarthic speech can be characterized by several particular dysfunctions caused by the loss of proper motor control. In the case of PD these include: dysphonia, imprecise articulation, dysprosody and speech volume intensity fluctuations.

Applications. Most of the studies focus on the detection of PD patients from the general healthy population [101, 102, 103, 104]. Other studies focus

on the differential diagnosis between PD, MSA and PSP [105], on progression monitoring [106] and severity monitoring [90]. Another promissing application for speech in PD analysis is the detection of PD in an incipient stage, before a clinical diagnostic is placed [104, 107]. Harel et al. [108] analyze speech of two English speaking PD patients and two age-matched healthy controls retrospectively and concluded that some of the frequency content analyzed was relevant for the early identification of PD.

Features used. When tackling dysarthic PD speech, statistical features are extracted from the time domain and specific frequency features are explored. There is a high number of proposed features as the field of general speech processing is well developed. The type of features selected depend on the type of problem studied, i.e., phonation, articulation, rhythm or volume. Time domain features include: duration of pause intervals, rate of speech timing, change in interval length, period of onset of vocalization, vowel keeping time, descripive statistical measures [109, 110, 111]. Frequency features are also diverse and some of the most relevant are: main frequency of vocal cord vibration or pitch, jitter, shimmer, noise-to-harmonics ratios, formant frequencies, vowel space area, pitch and amplitude perturbation quotient, Mel Frequency Cepstrum Coefficients [102, 112, 110, 111].

Classification Algorithms. For all classification problems tackling PD speech, i.e., PD vs. HC, differential diagnosis or severity monitoring, the most used classification algorithm is SVM [102, 110, 112]. The early work of Little et. al [102] used a SVM model with a Gaussian Radial Basis Kernel function and resulted in an accuracy for PD identification from speech of 91.4%. Arroyave et. al [110] distinguished between Spanish speaking PD patients and HC using a SVM with a soft margin and Gaussian kernel in a 10-fold cross validation strategy resulting also in a high accuracy of 91.3%. However, there are other relevant approaches. For instance, Mekyska et al. [106] use a Random Forest classifier for PD severity assessment on a database with 84 PD and 49 HC patients. A sensitivity of 92.86% was obtained. The early work of Das et al. [113] compares Neural Networks with Decision Trees and regression algorithms. The neural network provided the best result, with an accuracy of 92.9%.

Real-life Recordings. The majority of the studies looking at dysarthic PD speech focus on ideal voice recording conditions. However, PD detection algorithms should be sufficiently robust for real-life, noisy scenarios. Applications in early detection of PD would be most efficient in population screening scenarios. Correa et al. [103] analyze the effect of different noise conditions, e.g., saturation, dynamic compression, additive white Gaussian noise and different kinds of environmental noise. Results showed that different background environmental noises have a high impact on the classification results. Another method of continuously assessing the condition of PD patients is through the use of a mobile phone, either in a test application or through recording of phone conversations or via the cellular network. Rusz et al. [111] evaluates the use of smartphone speech recordings for early PD detection. The system was tested on 50 patients suffering from RBD and promising alterations in the speech pattern of prodromal PD subjects was obtained. The distinction between HC and

RBD patients was obtained with an AUC of 0.69, a sensitivity of 69.8% and a specificity of 64.7%.

4.3. Handwriting Analysis

The analysis of handwriting has proven effective in the diagnosis and progression monitoring of PD patients [18]. Handwriting is a complex activity involving both cognitive and motor functions. As the disease progresses and affects the brain centers responsible for its motor aspects, several abnormal characteristics of the handwriting activity can be observed. Micrographia, a reduction in the size of written text, is very often present in patients with PD. Bradykinesia and tremor also affect the ability of controlling the motions involved in writing. Such anomalies can be monitored either through *static* and/or *dynamic* approaches.

Static approaches. Refer to the graphical feature analysis of written text. The graphical characteristics are used to analyze the extent of micrographia and the randomness of strokes generated by tremor related movements. Typical metrics include changes in size of written characters, height of loop patterns, area of text blocks, pixel density variations based on ink content [114], density and height ratios, spiral precision index [115]. Besides providing an estimate of disease severity, these types of studies also allow for longitudinal tracking of PD progression and the identification of prodromal symptoms. For instance, Zhi et al. [114] explores the potential of using static analysis on historical signature based writing samples in the study of disease progression for 10 PD patients.

Dynamic approaches. Look at the kinematics of handwriting. In this case, symptoms related to bradykinesia, tremor and rigidity are assessed by also analyzing the on-surface and in-air movements associated with writing. Dynamic methods make use of digital tablets, smart pens with axial pressure of ink and tri-axial accelerometers [116] and EMG [115]. Depending on the modality of recording, different features are extracted. Digital tablets can usually record the point of contact (x and y directions) and pressure information. Several kinematic features are extracted including: speed of writing, changes in acceleration and velocity, writing duration and length, jerk, stroke length, descriptive statistical measures [117], the rate of pressure change with respect to time [118]. Smart pens in combination with digital tablets allow for additional tracking of in-air movements [116, 119]. Bradykinesia is assessed by calculating the movement time and velocity, whereas tremor by analyzing the frequency content of the pen tip trajectory during rest. The use of EMG was explored by Loconsole et. al [115] and specific signal features are extracted: root mean square, mean absolute value, zero crossings of the EMG signal.

Classification Algorithms. Machine learning algorithms are used for classification of the PD or HC states. Drotar et al. [118] use SVM in several handwriting classification tasks. This classification method obtained an accuracy of 81.3% on the kinematic and pressure features database PaHaW, composed of 37 PD patients and 38 HC [118]. Loconsole et al. [115] observed that SVM outperforms artificial neural networks with and without PCA based feature reduction on a smaller EMG database. Deep learning methods were also used by Pereira et al. [116]. The authors developed a convolutional neural network for

handwritten dynamics differentiation on the HandPD dataset comprised of 74 PD and 18 HC.

4.4. Face Video Analysis

In the process of PD neurodegeneration, neurons from the basal ganglia start dying leading to dysfunctions in the neuronal circuits controlling facial muscles. As a result, some PD patients suffer from hypomimia, a reduction in the facial muscle movements (facial bradykinesia).

Hypomimia in Parkinson's disease is quite a recent research topic and efforts are being made to better characterize these movement deficits. Gunnery et al. [120] used videos of participants mentioning pleasant activities to map spontaneous facial expressions in PD. The analysis was performed by extracting facial action units and characterizing features such as onset, offset and apex. Similarly, Livingstone et al. [121] used EMG to study facial muscle reaction during presentations of calm, happy, sad, angry and fearful emotions. Hypomimia was observed with a reduction in EMG amplitudes and delayed onset in the muscles controlling smiling. The video based analysis of facial expressions in PD patients relies on the general knowledge available for video facial emotion recognition and focuses on distinguishing healthy controls from diseased individuals. Bandini et al. [122] used a Multi-class SVM to train a facial expression recognition model from benchmarked databases. The test dataset comprised videos from both PD and HC. The performance of the model is proposed as an indication of hypomimia effects.

4.5. Brain Imaging

Some of the most common Parkinson's disease biomarkers are the changes observed in brain tissue through non-invasive imaging techniques. The identification and characterization of such biomarkers is important for placing an initial diagnostic and following disease progression. Perhaps the most relevant application is the use of brain imaging biomarkers for the differential diagnostic between PD and other neurodegenerative diseases with similar early symptomatology. Unlike the case of Alzheimer's disease, where brain imaging biomarkers have been extensively studied through computational approaches and specifically machine learning techniques [123], the automatic analysis of PD biomarkers is at an incipient stage.

Topics addressed. Brain imaging biomarkers are identified and characterized through the automatic analysis of MRI, SPECT or PET images with the purpose of differentiating between PD and HC [124, 125, 126, 127, 128], but also between PD and other neurodegenerative diseases (e.g., MSA [129, 130], PSP [131]). Haller et al. [124] used MRI diffusion tensor imaging with specific features extracted for the classification of PD vs. HC. For the early stage differentiation of PD from HC, MSA and PSP, Marquad et. al [130] studied the extraction of different anatomical features from the whole brain and a subcortical motor network with its component regions.

Features. The feature extraction process focuses not only on the type of information that could result in PD biomarkers, but also on the brain location

from where they are extracted [125, 126]. As PD affects different regions of the brain throughout different stages of progression, selecting the most probable brain area where the disease might manifest is relevant especially in early detection and differential diagnosis. Peng et al. [126] determined that the best classification results were obtained from the frontal, parietal, limbic and temporal lobes and the central region. With regard to the type of features, these can be voxel based morphometric features or low level features related to the volume of grey matter, white matter and cerebral spinal fluid, but also high level features that represent the structural connectivity [126]. Singh et al. [128] proposed an unsupervised feature extraction method In combination with a least square SVM, PD data was distinguished from HC data with a 99% accuracy in a hold out validation procedure.

Classification Algorithms. The most used algorithm for all the classification problems is SVM with different implementations [131, 125, 126]. Adeli et al. [125] obtained the best performance with a LDA classifier in combination with a joint feature sample selection for PD detection on the PPMI database. Hirschauer et al. [127] build an enhanced probabilistic neural network (EPNN) with four layers for classifying PD patients with respect to HC. The use of EPNN resulted in a classification accuracy of 92%.

4.6. Multimedia Approaches

The symptoms affecting PD patients can be diverse and monitoring only one of them might be insufficient for providing a good estimate of the disease progression. Using multiple modalities for assessing the patient's state could be beneficial. These can also be integrated in the day to day activities of the patient, not only to monitor the disease progress, but also the effectiveness of the treatment and adherence to medication.

Mobile Applications. The HopkinsPD is an application proposed by Zhan et. al [132] that aims to remotely monitor PD symptoms through a smartphone platform. Five symptom types are analyzed: voice dysphonia, postural instability while standing up, gait - bradykinesia, reduced dexterity and rest tremor. Data is collected from the phone microphone, accelerometer sensor, push of a button and different self evaluation questionnaires. The study was deployed world wide through a mobile application and recorded data from 221 PD and 105 HC. An accuracy of 71% was reached. A similar approach was implemented by Neto et al. [133] by using iPhone sensor data for medication response detection. The best performing classification algorithm on the specific features extracted were tree based tests, including random forest. In the work of Adams et al. [134], keystrokes recorded with the App-Tappy application were used for classifying early PD and HC. The features extracted included hold time, statistic measures, latency measures and statistics on latency. Several machine learning algorithms were tested.

Other smartphone applications focus on assessing dexterity of PD patients [135, 136]. Aghanavesi et. al [136] used a smartphone to track how subjects performed tapping and spiral drawing tests. Several features were extracted and pre-processed using PCA. The best result for predicting PD symptom severity

Purpose	Modality	Dataset	Classifier	Eval	Acc	\mathbf{Ref}
PD vs. HC	ground reaction force	29PD, 18HC	SVM	LOO	93.6%	[74]
PD vs. HC	ground reaction force	93PD, 73HC	LSTM, CNN	hold out	98.7%	[83]
PD vs. HC	speech data	50PD, 50HC	SVM	10fold CV	91.3%	[110]
PD vs. HC	handwriting dynamics	37PD, 38HC	SVM	10fold CV	81.3%	[118]
PD vs. HC	handwriting dynamics	74PD, 18HC	CNN	hold out	95%	[116]
PD vs. HC	MRI	518PD, 245HC	SVM	CV	99%	[128]
PD vs. HC	MRI	200PD, 375HC	EPNN	hold out	92%	[127]
PD vs. HC	speech data, hand- written dynamics, gait signals	44PD, 41HC	CNN, SVM	hold out	97.6%	[138]
symptom severity	accelerometer	12PD	SVM	LOO	90%	[84]
severity of tremor	accelerometer	18PD, 5HC	HMM	LOO	82%	[93]
detect FoG	inertial sensors	21PD	CNN	hold out	90%	[87]
detect dyskinesia	accelerometer	5 HC, $14 PDwith D, 10 PD$	ANN	LOO	84.3%	[96]
medication effect	speech data, ac- celerometer, push of a button, question- naires	221PC, 105HC	RF	10fold CV	71%	[132]

Table 4: Overview of the most relevant research works using machine learning in handling Parkinson's disease that were presented in this work. Brief details are provided on the dataset size and content, classification techniques and evaluation methods.

was obtained using a SVM classifier which resulted in a high correlation with the UPDRS ratings of each participant.

Speech and writing. Afonso et. al [137] have used deep learning methods for assessing PD based on voice analysis and dynamic techniques for writing assessment. A deep optimum-path forest clustering technique was used on 31 PD and 35 HC performing hand movements and drawing with a biometric pen incorporating a michrophone, finger grip, axial pressure of ink, tilt and acceleration. In the work of Correa et. al [138], speech, handwriting and gait signals are analyzed in a database containing 44 PD patients and 40 HC. A CNN is used for the multimodal analysis of PD patient data. The features obtained from the last hidden layer of the CNN are placed into a subject specific feature vector and fed to an SVM classifier.

4.7. Summary

An abundance of literature is available on the characterization of PD motor symptoms with computational approaches. The most focus is in the area of altered movement patterns. Speech disorders caused by PD have also been often investigated as speech is easy to record and the field of voice analysis has significant history. A summary of the most prominent works using machine learning for PD characterization presented in this literature review is available in Table 4.

5. Huntington's Diseases

This section presents a review of the prominent computational approaches used in the diagnosis and monitoring of Huntington's disease. Its prevalence is significantly lower than in the case of AD and PD. As a consequence, the amount of studies conducted with the purpose of developing computational approaches for its monitoring and diagnosis is significantly smaller. The gait of HD patients is characterized by uncontrolled, hyperkinetic movements such as *chorea* and *dyskinesia*. Automatic monitoring of motor symptoms can be useful in analyzing disease progression.

5.1. Gait Abnormalities

Classification of HD. The gait of HD patients presents significant differences from that of HC and these differences are still a subject of research. In the study presented by Pyo et. al [139], the step length, stride length and base support and their corresponding coefficients of variation of HD patients proved to be increased when compared to HC. Mirek et. al [140] used magnetic trackers to calculate the gait cycles. Results show the HD patients present insufficient flexion in the plantar and knee joints and excessive flexion of the hip when compared to normal gait parameters. Automated classification of HD gait signals has also been investigated. Manini et. al [141] uses inertial sensors attached to the ankles and the lumbar region to record stance and swing in 10 HD patients rated according to the UHDRS scale (Unified Huntington's Disease Rating Scale - clinical assessment of HD severity), 10 post-stroke patients and 10 HC. A HMM was trained in a supervised way to recognize the foot strike and toe off events with a delay of 20ms.

Differential Diagnosis. Mann et. al [142] also used a magnetic tracker for analyzing the motion of arms of PD and HD subjects. The movements caused by the two neurological disorders and recorded with the magnetic trackers were characterized by their amplitude, frequency, dispersion, entropy and other statistical features. By studying the subtle differences in abnormal movements, a more accurate initial diagnosis can be provided. For instance, Dinesh et al. [143] place a wearable sensor (BioStampRC) on the arms and legs of 10 PD, 10 HC and 15 HC for motion characterization in a simple walking test. The signals recorded included 3D accelerometers, ECG and EMG. The features extracted showed a good visual discrimination between the three conditions.

5.2. Speech Impairments

Basal ganglia neurodegeneration leads to motor impairments which might affect the muscles involved in speech production. Different diseases might cause different types of abnormalities in muscle control and hence in the produced speech. Differentiation between HD and other neurodegenerative diseases such as PD, MSA and PSP, based on voice recordings, was proposed by Rusz et. al [144]. Repetitions of the 'pa' syllable where characterized by features representing rhythm instability and acceleration through the detection of syllable onset. The accuracy of the syllable onset detector was of 99.6% and the visual observation of the features showed discrimination power between syndromes. Novotny et. al [145] characterized PD and HD dysarthia by also looking at syllable onset in 'pa-ta-ka' repetitions using the Hilbert transform. The accuracy of PD syllable onset detection was of 90%, while for HD it was 80%.

5.3. Biomarkers

5.3.1. Brain Imaging

Brain imaging is considered one of the most reliable methods used for confirming a HD diagnosis. Several works have focused on the development of (i) MRI biomarkers for the characterization of HD and their presence (ii) prior to disease onset.

Classification of HD. Rizk-Jackson et al. [146] uses a database of MRIs of 39 HD and 25 HC patients to extract region-based and voxel-based features from white and grey matter. Using an LDA classifier a balanced accuracy of 76% was obtained for differentiating HD and HC.

Pre-onset HD detection. Although HD is a genetic disease and carriers of the huntingtin gene are already aware they will develop the disease, the exact onset is not yet predictable. Several studies have tried to identify pre-HD signs through brain imaging biomarkers several years prior to disease onset. For HD carriers, an MRI scan is typically taken every 2 years. In practice, that is not always the case. Eirola et. al [147] proposed an extreme learning machine with a hidden layer of 1,000 neurons for predicting the onset of HD 10 years in advance. The output result showed an accuracy of 80-90% over the entire 10 year period. In the Predict-HD study, the MRI scans from a total of 95 preHD subjects and 95 HC subjects were used for predicting HD several years before onset. Information was extracted from the gray matter of several regions of interest and fed to a multivariate SVM. By selecting the region of interest, an accuracy of 83% was obtained. The performance of the classification of the preHD subjects increased as time to onset decreased. Mason et. al [148] used the Track HD consortium data with MRI scans from 19 preHD and 21 HC subjects to extract both structural and connectivity measures. Using a linear support vector machine preHD was identified 5 years prior to disease onset and a maximum accuracy of 88% was obtained. DTI was also used by Karistianis et al. [149] in obtaining biomarkers for the preHD versus HC discrimination. Different tests were performed for extracting features either from the whole brain or from specific regions. A quadratic discriminant analysis showed a good discrimination power for the volumetric reduction and increased fractional anisotropy in the basal ganglia up to 15 years prior to onset.

5.3.2. EEG Signal

As Huntington's disease also implies cognitive and psychological impairments, changes in the activity of the brain might also be observed in the EEG

-						
Purpose	Modality	Dataset	Classifier	Eval	Acc	\mathbf{Ref}
HD vs. HC	EEG	26HD, 25HC	SVM	10 fold CV	83%	[151]
HD vs. HC	MRI	39HD, 25HC	LDA	4fold CV	76%	[146]
HD onset	MRI	1370HD	ANN	hold out	90%	[147]
HD onset	MRI	19preHD, 21HC	SVM	LOO	88%	[148]
discriminate stance and swing	inertial sen- sors	10HD, 10HC	HMM	-	20ms error	[141]

Table 5: Overview of the most relevant research works using machine learning in handling Huntington's disease that were described in this work. Brief details are provided on the dataset size and content, classification techniques and evaluation methods.

measurements. Tommaso et. al [150] analyzed recordings from 13 HD in order to extract possible EEG biomarkers. Features extracted from the Fourier transform of EEG signals were fed into an artificial neural network classifier which correctly predicted 11 out of 13 subjects as containing the HD gene. Odish et. al [151] went further to create an EEG index on 2 seconds Fourier transformed data. The proposed method was tested by selecting the adequate features through PCA and training an SVM model. The classification was tested on 26 HD gene carriers and 25 HC resulting in an accuracy of 83%.

5.4. Summary

As Huntington's disease is less prevalent in the population and has a strong genetic correlation, fewer research works investigating HD were found when compared to publications investigating AD or PD. Machine learning techniques are used in the study of HD to bring more clarity on the onset of the disease. A summary of the most prominent works presented in this work using machine learning for HD is available in Table 5.

6. Amyotrophic Lateral Sclerosis

Amyotrophic Lateral Sclerosis is a severe motor neurodegenerative disease with a rapid progression and a low prevalance when compared to other neurodegenerative diseases. Studies that rely on computational approaches focus mostly on providing an aid to ALS diagnosis, most frequently looking at a better differentiation from other neurodegenerative diseases at incipient phases with predominant motor abnormalities such as PD, HD or MSA.

6.1. Gait Abnormalities

Characterization of gait in neurodegenerative diseases, particularly in ALS, HD and PD, has been performed by Hausdorff et. al [152] by recording the magnitude, duration of stride-to-stride fluctuations and perturbations in the fluctuations dynamics by using force sensors placed on the feet of the subjects. The gait of ALS patients was less steady and more temporarily disorganized. No

other studies using other databases (private or public) characterizing gait dynamics of ALS patients were found by the authors. Using the neurodegenerative diseases gait dynamics database, Dutta et. al [153] automatically identified the different disorders from healthy controls using several features extracted from a cross-correlogram with an Elman's recurrent neural network with one hidden layer. The result for binary classification (ALS vs. HC) was in the range of 90.6% to 97.8% average accuracy. Xia et. al [154] used the Teager Energy Operator to extract features for an SVM classifier. The proposed method resulted in an accuracy of 92.86%.

6.2. Speech Impairments

The muscles involving speech production are also affected in any otrophic lateral sclerosis. ALS speech abnormalities were investigated by Yunusova et. al [155] through features as articulatory rate, duration of speech and pauses. The aim of the study was to evaluate the effect on speech of diseases predominantly characterized by motor deficits and those with a predominant cognitive deficit. In this case, significant statistical differences were observed between features extracted from the speech of ALS subjects when compared to the speech of fronto-temporal dementia. Wang et. al [156] investigated the possibilities of automatically detecting ALS from speech. The proposed method used both acoustic information and articulatory movement data. The movement data was recorded with electromagnetic articulograph sensors attached to the tongue and lips of the participant. Features were extracted from both signal sources and two classifiers were compared: SVM and a DNN. The SVM was able to classify ALS speech with an accuracy of 80.91% using both acoustic and movement data. The DNN resulted in 91.74% accuracy only using acoustic data in a 4-fold cross-validation scenario.

6.3. Video Analysis

For the purposes of early diagnosis and tracking of ALS, Bandini et. al [157] investigated the use of kinematic features extracted from videos of the face while ALS and HC subjects were performing both speech and non-speech tasks. As ALS affects all motor neurons, the muscles of the face are also impaired. Using a logistic regression classifier, an overall accuracy of 88.9% was obtained when discriminating between ALS and HC.

6.4. Biomarkers

6.4.1. Brain Imaging

It is used in ALS to investigate its causes and progression. Fekete et. al [158] used MRI brain scans from 40 ALS and 30 HC subjects to propose an ALS biomarker based on the organization of brain networks at a functional level. Features were extracted from the 0.03-0.06 Hz band using a typical image processing chain: motion correction was applied, followed by a normalization to the MNI (Montreal Neurological Institute) space and the use of masks for CSF (cerebrospinal fluid) and white matter extraction. For the classification task,

Table 6: Overview of the most relevant research works using machine learning in handling ALS that were presented in this work.Brief details are provided on the dataset size and content, classification techniques and evaluation methods.

Purpose	Modality	Dataset	Classifier	Eval	Acc	Ref
ALS vs. HC	force sensors	13ALS, 16HC	RNN	hold out	97.8%	[153]
ALS vs. HC	force sensors	13ALS, 16HC	SVM	LOO	92.86%	[154]
ALS vs. HC	speech data, ar- ticulatory move- ment data	11ALS, 11HC	DNN	LOO	91.74%	[156]
ALS vs. HC	video data	10ALS	Regression	LOO	88.9%	[157]
ALS vs. HC	MRI	40ALS, 30HC	SVM	LOO	87%	[158]
ALS vs. HC	EMG	10ALS, 11HC	LDA	LOO	90% sen- sitivity	[160]
survival of patients	MRI	135ALS	DNN	hold out	84.4%	[159]

an SVM with a recursive kernel elimination was used leading to an accuracy, sensitivity and specificity of 87%, 88% and 88%, respectively. A method for predicting the survival in ALS patients was proposed by Van der Burgh et. al [159]. Both MRI and clinical characteristics were studied on 135 ALS patients classified as short, medium or long-term survivors. A deep neural network was used for prediction, leading to an accuracy of 84.4%.

6.4.2. EMG Signal

As motor muscles are significantly affected in ALS, using non-invasive EMG measurements for ALS diagnosis could be a cost effective option for an initial diagnosis. Zhang et. al [160] investigated several statistical features extracted from the EMG of ALS patients and HC subjects. These were used with an LDA classifier. The classification provided a sensitivity of 90% and a specificity of 100% for differentiating ALS and HC subjects from EMG data.

6.5. Summary

The characterization of ALS through computational methods is limited in the available literature. The study of the disease through machine learning techniques is still at an early stage. Most research works focus on identifyin ALS patients from HC. A summary of the research works using machine learning on ALS that are available in this literature review are presented in Table 6.

7. Multiple System Atrophy

Multiple System Atrophy is a severe neurodegenerative disease that progresses rapidly after onset. Its prevalence is small and as a consequence the number of studies using computational approaches for its characterization is small. Most of these studies focus on providing better methods of diagnosis in the incipient stages of the disease when the symptoms are confusing. MSA bares the closest resemblance to Parkinson's disease, especially the parkinsonian version of MSA (MSA-P). Computational approaches can help in its better diagnosis by analyzing imaging biomarkers, evaluating speech alterations and analysis of proteomics data.

7.1. Biomarkers

According to Duchesne et. al [129], the distinction of Parkinson's disease and parkinsonian plus syndromes presents an initial error rate of up to 35%. Brain imaging is generally useful in providing more insights into the correct diagnosis. Developing biomarkers for automatic detection has been studied by Duchesne et. al [129] on MRI data from MSA and PD patients. After image pre-processing, the tissue composition and deformations from the hind brain were evaluated for their discriminative power. A model created on these features with an SVM least square optimization algorithm provided an accuracy, specificity and sensitivity of 91%, 88% or 93% respectively. Similarly, in the study of Marquand et. al [130] the midbrain was the anatomical region with most discriminative power for the selection of nerodegenerative biomarkers. The MRI images were collected from PSP, PD and MSA patients. An SVM model was created, leading to an accuracy of 91.7% for MSA detection.

7.2. Speech impairments

Due to the degeneration of neurons in the basal ganglia, control of the muscles producing speech might be affected in MSA, particularly in the parkinsonian variant of the disease. MSA-P presents similar symptoms to PD. The study of Eun et. al [161] analyzes the differences in speech patterns between the two diseases. Subjects suffering from MSA-P showed more speech impairments than those with PD, reflected in the voice pitch, prolonged pause time and reduced speech rate. Soler et. al [162] proposed an analysis of the fundamental frequency of snoring for the identification of MSA patients. Although a slightly different approach than that of Eun et al. [161], the method analyzes the sounds produced by the same muscles affected by neurodegeneration.

7.3. Summary

Very few studies looking at MSA with computational approaches were found. Machine learning techniques are used to help in the differential diagnosis between MSA and other similar motor disorders. A summary of the studies using machine learning for MSA from this literature review is available in Table 7.

8. Sleep disorders present in Various Diseases

This section presents an overview of studies analyzing abnormal sleep behaviors manifested in several neurodegenerative diseases. More specifically, we investigate the use of computational approaches used for: REM sleep behavior disorder and Periodic Leg Movements (with or without restless legs syndrome).

Table 7: Overview of the most relevant research works using machine learning in handling ALS. Brief details are provided on the dataset size and content, classification techniques and evaluation methods.

Purpose	Modality	Dataset	Classifier	Eval	Acc	Ref
MSA vs. PD	MRI	$16\mathrm{PSP}$ or MSA, $16\mathrm{PD}$	SVM	LOO	91%	[129]
MSA vs. PD, PSP	MRI	$12\mathrm{PSP},14\mathrm{PD},19\mathrm{MSA}$	SVM	LOO	91.7%	[130]

These symptoms are presented separately as they are present in multiple diseases characterized by alpha-synucleinopathies such as PD, MSA or DLB [163] and have similar clinical characteristics throughout all diseases. RBD and PLM have been until recently characterized as separate disorders. In recent years, the link with neurodegeneration has been firmly established. RBD is now considered as part of alpha-synucleinopathic degeneration and its appearance years prior to the disease onset is a prodromal symptom [164].

8.1. REM Sleep Behavior Disorder

Rapid eye movement sleep behavior disorder is characterized by the enactment of dreams and unusual motor behavior during REM sleep, more precisely REM sleep without atonia (RSWA). The prevalance of RBD differs per type of neurodegenerative disorder. In MSA and DLB, more than 80% of patients develop RBD. For PD, the number of patients who present RBD symptoms is lower. Computational methods are used both for the characterization of RBD from polysomnographic recordings but also for the automatic detection of the abnormal recordings related to RBD.

Characterization of RBD. In recent years, changes in EEG activity in individuals with RBD have been identified. Brazete et. al [165] have showed that RBD is linked to a slowing down of EEG activity during wakefulness, with delta and theta bands presenting higher spectral powers in RBD patients versus HC. Ruffini et. al [166] also investigated EEG complexity during awake EEG recordings using Lempel-Ziv-Welch Compression Spectrograms and entropy measures . RBD is present in the prodromal phases of multiple neurodegenerative diseases. Berrada et. al [167] attempted to differentiate between RBD patients who later on develop DLB and patients who develop PD from polysomnographic recordings and data extracted from clinical, neurological and neurophysiological exams. By applying an alternating decision tree, the algorithm was able to automatically differentiate only between RBD and HC subjects, with no significant results on the DLB or PD conversion.

Automatic Detection. The automatic detection of RBD patients from HC subjects is of interest for the reduction in time required for placing a diagnostic. Several threshold EMG based methods have been proposed [168, 169]. Cesari et. al [169] compare several available threshold-based methods. The Frandsen Index method outperformed the others with average sensitivity values of 90% for RBD patient detection. Kempfner et. al [170] proposed an RBD detection method based on the entire polysomnographic recording. Subject specific

features were extracted from all signals and were used as input to an SVM classifier. The area under the curve in a leave-one-out testing scenario was 0f 0.988 when using all signals and 0.981 when using only EMG activity. Ruffini et. al [171] proposed an automatic detection method for RBD based on awake EEG recordings. Two classification algorithms were proposed and compared: a DCNN with a 5-layer architecture and a Recurrent Neural Network with three stacked LSTM cells. The DCNN provided the best results with a classification accuracy of 80% between RBD patients and HC in a leave-one-out validation scenario.

8.2. Restless Legs Syndrome and Periodic Limb Movement

Periodic limb movements appear in patients with and without Restless leg syndrome and are characterized by uncontrolled limb movements that occur during sleep [172]. RLS and PLM are present in movement related neurodegenerative disorders. The highest prevalence is in PD patients. Due to a paucity of studies, it is difficult to establish the prevalence of these movement disorders in other neurodegenerative diseases. Their presence has been observed also in MSA, PSP and HD [172]. Similar to RBD, the majority of studies including computational methods focus on the characterization of RLS and PLM or on their automatic detection either from polysomnographic or actigraphy recordings.

Characterization. The characterization of PLM through EMG recordings during sleep is still a topic of investigation. Different states of RLS and PLM were investigated by Ferri et. al [173] by introducing a periodicity index and using Markov chains for the characterization of the structure of leg movement sequences. Ferrillo et. al [174] investigated the awakenings and EEG arousal prior and after PLM events. By analyzing the content of EEG signals through the wavelet transform and extracting the heart rate from cardiac signals, a significant increase in the heart rate and delta activity power was observed 3 to 4 seconds prior to PLM onset. Similarly, Sieminski et. al [175] looked at the spectral powers of the alpha, beta and delta bands of the EEG and found an arousal after PLM activity was detected.

Automatic Detection. The automatic detection of PLM events could reduce the time required for manual polysomnographic annotations. Tkach et. al [176] investigated the stability of time-domain features extracted from EMGs recorded from several muscles during PLM for their automatic detection. Using several statistical features and a linear discriminant analysis classifier, an improvement of 16% was obtained by feature and recording site selection. On the other side, Umut et. al [177] investigated the automatic detection of PLM using all PSG signals except EMG. A combination of Haar wavelet decomposition and Discrete Fourier Transform was used to extract the power from the delta, theta, alpha and beta EEG frequency bands. A kNN classifier used in a 10-fold cross validation scheme provided the highest accuracy of 91.87% for the detection of PLM events. The use of limb actigraphy might be a good alternative to PSG for classifying sleep disorders. Several commercial actigraphy devices are already available and have been tested for their utility in PLM detection. These

Table 8: Overview of the most relevant research works using machine learning in handling Sleep Disorders associated to neurodegenerative diseases. Brief details are provided on the dataset size and content, classification techniques and evaluation methods.

Purpose	Modality	Dataset	Classifier	Eval	Acc	Ref
RBD vs. HC	PSG	16RBD, 16HC	SVM	LOO	AUC 0.988	[170]
RBD vs. HC	EEG	121RBD, 91HC	DCNN	LOO	80%	[171]
detect PLM	PSG	153PLM	kNN	10fold CV	91.87%	[177]

include PAM-RL [178], the CE marked actigraphy device KickStrip [179] and Respironics Actigraphy [180].

8.3. Summary

Most literature available for sleep disorders related to neurodegeneration, such as PLM or RBD, propose different methods for their characterization. A few studies use machine learning for their automatic detection from regular sleep or for differentiating patients from HC. The studies presented in this literature review using machine learning for sleep disorders are summarized in Table 8.

9. Overview of the Common Processing Steps

After reviewing the relevant literature for the various neurodegenerative disseases, in this section, we identify and analyze the common processing steps employed by the computational algorithms and machine learning techniques. They are depicted in Figure 3. We propose a view that divides the classification process into six blocks. Neurodegenerative diseases and their symptoms are diverse and so the types of datasets available for different classification problems are varied. However, regardless of the data types, similar steps follow in case of classification problems. The raw data can be directly fed into the classification algorithm or several pre-processing steps are applied prior to classification. In some cases, the data might be pre-processed which can imply filtering, normalization or dimensionality reduction. For some studies, different types of features are extracted and further selected based on specific relevance measures. Regardless of the type of pre-processing applied, a classification algorithm is applied and the result is evaluated.

We detail these aspects in the following: a summary of the datasets available for training computational methods (see Section 9.1), an overview of the preprocessing, feature extraction and feature selection methods (see section 9.2), an analysis of the classification algorithms (see section 9.3) and evaluation methods (see section 9.4 found in this literature review.

9.1. Datasets

The spectrum of neurodegenerative disorders affects patients in different ways resulting in a variety of symptoms. The type of signals, protocols and



Figure 3: Overview of the general steps used in different classification problems for neurod-generative diseases.

information required to accurately diagnose or monitor these diseases are diverse. Therefore the datasets found in the literature proposing computational approaches show a mixture of recorded data, protocol for recording and size. The chosen protocol depends on the end goal of the study, e.g., to aid in the diagnosis of the disease, to monitor progression, to help in the differential diagnosis, to detect prodromal stages of the disease etc. Most of the times, the data collected is disease specific and can not be used for other diseases as it was not collected with a protocol developed for that purpose.

A high number of studies record specific datasets for their chosen topic of study such as a small variation in symptoms or different sensor recording modalities. Therefore the amount of data collected is most of the times small, from a few participants. The majority of the datasets used are small (67.5 %), with less than 50 participants per class. These datasets tend to be private and target specific diseases or symptoms. Medium sized datasets (between 50 and 100 participants per class) make up 14.41% of the datasets considered in this study and large datasets (with more than 100 participants per class) make up 18.07%. Large and medium size datasets are predominant in all diseases in topics such as Speech processing, Brain Image analysis, Classification of Tests and Medical Records or Genetic information. Big sized datasets are generally collected in consortium or projects spanning multiple years. The majority of large datasets are made available to the research community free of charge creating the opportunity for more researchers to work on the development of adequate solutions. In Table 9, an overview of the most relevant datasets is provided along with details on their content, availability and size.

9.2. Pre-processing, Feature Extraction and Selection

In some of the studies, when predicting a certain condition or using computational approaches for placing a diagnostic, features are extracted from the raw data recorded and fed into a classification algorithm. These features are diverse and are strongly deendent on the type of input data. Prior to using features as input to a classifier, it might be useful to select relevant features or to project these into a more representative space while reducing the dataset dimensions.

When looking at feature selection, most of the studies make use of statistical measures for eliminating correlated features which add no or little additional information to the dataset. Statistical methods include computing correlation

Name	Disease	• Year	Avail.	Purpose	Data Type	Size
<i>ADNI</i> [181]	AD	2004- 2016	open	detection of AD and pre- AD; monitoring of pro- gression	clinical, genetic, MRI, PET, biomarkers	$ \begin{array}{llllllllllllllllllllllllllllllllllll$
AZTIAHO [62]	$^{\rm AD}$	2013	closed	speech biomarkers for AD	speech data	50 HC, 20 AD
CADDementia [182]	$^{\mathrm{AD}}$	2015	open	detection of HC, MCI and AD	MRI data	384 recordings
Daphnet [86]	PD	2008	open	freezing of gait	accelerometer data	10 subjects
DementiaBank [183]	AD	1987- 2019	open	speech biomarkers of de- mentia	speech data	167 AD and 97 HC
Gait in Neu- rodegenerative Disease [82]	$_{ m HD,}^{ m PD,}$	2000	open	gait dynamics and re- sponse to medication	force sensor recordings	15 PD, 20 HD, 13 ALS, and 16 HC
Gait in Aging and Disease [82]	PD	1997	open	gait in PD	force sensor recordings	5 PD, 5 old, and 5 HC
Gait in Parkin- son's Disease [184]	PD	2005	open	gait in PD	force sensor recordings	93 PD, 73 HC
Oxford PD Detec- tion [102]	PD	2009	open	dysphonia in PD	speech data	23 PD, 8 HC
OASIS [185]	$^{\mathrm{AD}}$	2007- 2010	open	detection of AD	MRI, PET	OASIS1, 416, ASIS2, 150, and OASIS3, 1098 subjects
PaHaW [186]	PD	2016	open	archiedian spiral draw- ings and handwriting for PD	ressure, xy-coordinates, tilt, elevation, and in-air/on-air surface status	37 PD, 38 HC
PC-GITA [101]	PD	2014	closed	speech in PD	speech data	50 PD, 50 HC
$PD \ Speech \ [187]$	PD	2014	open	speech in PD	speech data	20 PD, 20 HC
PD spiral draw- ings [188]	РD	2013	open	drawings in PD	digital tablet parameters	62 PD, 15 HC
PDMultiMC [189]	PD	2017	closed	handwriting, speech and eye movements in PD	speech, digital tablet param- eters	16 PD, 16 HC
PPMI [190]	PD	2002	open	biomarkers for PD	brain images, clinical data, biological samples	432 PD, 196 HC, 64 early-PD, 65 RBD
Predict HD [191]	НD	2008	closed	HD detection correlated to genetic data	genetic data, cognitive as- sessemnt, tapping test, ver- bal learning/memory task, odour recognition, MRI	438 pre-HD
Track HD [192]	НD	2008- 2014	closed	HD detection correlated to genetic data	genetic data, MRI, clinical, cognitive, quantitative mo- tor, oculomotor and neu- ropsychiatric assessments	366 participants
Tickle- Degen [193]	PD	2010	closed	quality of life in PD	video recordings	117 PD

Table 9: Selection of the most relevant datasets found in literature .

coefficients, t-tests, Whitney U-tests, Kruskal Wallis tests or mutual information [61] [194]. Other methods are based on entropy or information gain [195].

More complex feature selection techniques such as forward feature selection are also used. This technique adds features one by one as input to the classifier and selects the ones that improve the classification performance [74][196]. It is not recommended for high volume datasets. Other methods for feature selection use regression techniques such as LASSO (least absolute shrinkage and selection operator) or the feature importance computed using the Random Forest algorithm [112]. Tsanas et. al [112] also uses the RELIEF feature selection algorithm that also considers the interaction between the different features.

Dimensionality reduction is also employed in some of the studies as a preprocessing step. The most popular methods are factor analysis [58], principal component analysis [129, 151], independent component analysis [197] and autoencoders [70, 45]. Besides reducing the dimension of the input data set, the information is projected in different dimensions that might enhance the classification performance.

9.3. Classification Algorithms

The problems approached can be binary, such as looking at whether a disease is present or not, or divided into multiple classes, when differentiating between different diseases or different stages of progression of a disease. The methods used for classifying the targeted states can be as simple as using threshold-based algorithms or imply the use of advanced machine learning methods.

Out of the studies considered in this review approximately 64% use machine learning algorithms. Figure 4 provides an overview of the types of algorithms used in classification problems. A typical processing chain involves the preprocessing of raw signals followed by feature extraction and classification. Most proposed methods make use of supervised learning techniques where a labeled training set is presented to the classifier for building a model. Such techniques include Linear Regression, Naive Bayes, SVM, k-NN, Random Forests, Decision Trees, LDA. By far the most used classifier in all researched diseases is SVM. The popularity of SVM can also be explained by the problems tackled: a big majority of the studies look at identifying between patients and healthy controls and so handling binary classification problems. Shallow and deep neural networks have also been used, but they are not as popular as the more conventional algorithms enumerated before. Some studies use different combinations and variations of MLP, CNN, DNN, ANN. Other types of classification algorithms look into more elaborated network architectures such as EasyMKLFS, Extreme Learning, Probabilistic Neural Networks, Gaussian Neural Networks and Deep Belief Networks. Unsupervised learning was also attempted in more recent years via autoencoder networks. Deep neural networks have been more commonly used in studies using images as input for analysis, and are gaining more and more traction nowadays.



Figure 4: Overview of the employed classification algorithms (as analyzed in this review) represented on a per disease basis. The most encountered algorithms include: Th-b - threshold based, LR - Linear Regression, NB - Naive Bayes, SVM - Support Vector Machines, kNN - k-nearest Neighbours, RF - Random Forests, DT - Decision Trees, LDA - Linear Discriminant Analysis, MLP - Multilayer Perceptron, CNN - Convolutional Neural Networks, ANN - Artificial Neural Networks.

9.4. Evaluation Metrics

We overview the common practices for assessing the performance of the computational systems. This brings into discussion the way the data is used for training the systems and the metrics employed for assessing the actual performance.

9.4.1. Data splitting

Although the commonly employed practices for training and validating the systems are the ones used in machine learning, there are some adaptations to the specificity of the data. We overview here the common practices: k-fold cross-validation — It tests the performance of the model on different unseen portions of the same type of data. The entire available data set is split intro k-folds of equal size. From this division, k-1 folds are used for the training and the kth fold for testing. Besides providing an indication of how the model would react to unseen data, it can also be useful when handling large amounts of data as the data is fed into the training and test phases in folds and not through large blocks: *leave-one out* - It is typically used with small datasets when using k-fold cross-validation would significantly reduce the amount of data available for training. In this case, the data from one subject is kept for testing, while the rest of the subjects are used for training. The testing and training sets are rotated until all subjects have been used for testing. Another variation of the leave-one-out validation is the leave-one-record out, where instead of using subjects, only one record (unit of the data set) is used for testing while the others are used in training. This technique is more resource consuming than kfold cross-validation and is generally not recommended for large datasets; hold out - Part of the data set is kept for training while the other for testing. A

typical division would be 80% training and 20% testing, but variations exist. By dividing the data set, this validation allows the evaluation of performance on completely unseen data. However, the single division into test and train sets can lead to a local optimum result, which would not generalize well to a real-world scenario.

9.4.2. Metrics

Among the exiting metrics employed in machine learning and information retrieval, some are more predominantly found when dealing with data from neurodegenerative diseases. We present here the most prevalent ones: accuracy represents the ratio between the total number of correct predictions and the total number of predictions. It does not provide insights into the rate of true positive and true negative predictions while also ignoring per-class performance evaluation. An algorithm for differentiation between different diseases can have a high predictive power for one disease and an extremely low one for a different diseases. However, the overall accuracy would be at an acceptable level; specificity (true negative rate) — represents the number of true negatives from the total number of predictions that are correctly identified; sensitivity (recall or true positive rate) — represents the number of true positives over the total number of predictions. It gives an indication of how well the algorithm detects specific classes; Area Under the Curve (AUC) — it is a method suitable for evaluating multi-class problems. It estimates the area under the receiver operating characteristic curve (ROC) which relates the true positive to the true negative rates at different settings of the classifier; F1-Score — represents a harmonic mean between precision (positive predictive value) and recall (probability of detection); Cohen's Kappa coefficient — is a statistical method that is typically used to quantify inter-rater agreement. The Kappa coefficient is computed between test labels and the predicted values obtained as output from the classification; Correlation coefficient — provides a measure of the strength of similarity between the prediction result and the desired output.

10. Conclusions and Future Challenges

In this review, we provided an overview of the general trends in employing computational approaches for the monitoring and diagnosis of neurodegenerative diseases. We have focused our efforts on five neurodegenerative diseases representative for the entire spectrum of neurodegeneration: AD, PD, HD, MSA and ALS. Neurodegenerative diseases have been extensively studied in recent years with the help of computational approaches, especially via traditional machine learning or deep learning networks. Diseases that present a higher occurrence rate, such as Alzheimer's and Parkinson's disease, are more often investigated. The higher economic burden imposed by more prevalent diseases has pushed for faster results and led to more solutions for automatic diagnostic and health assessment systems. Many solutions propose methods that incorporate classification algorithms.

10.1. Current State of Research

Datasets and Evaluation. Many small, closed datasets tailored to specific diseases and symptoms are used in the existing research. Surprisingly, larger datasets are mostly public. However, the available annotated data is not yet capable of coping with the actual requirements of deep neural networks, to allow maximum performance. Another aspect is the large variation in proposed evaluation techniques and metrics, which makes results difficult to compare, even on the same data set. Usually, the same data set is used for training and testing with different data folds. Testing on different collections than the ones used for learning is not as common. There is no consensus on evaluation metrics. Confusion matrices are rarely used even when dealing with multi-class problems. Disease specific scales are used as a golden standard only for PD, e.g., UPDRS and H&Y, but similar scales are rarely used as a reference for the other neurodegenerative diseases.

Challenges in Computational Approaches. Most computational methods proposed for the study of neurodegenerative diseases make use of shallow networks and handcrafted features. Deep learning networks along with the extraction of features in an unsupervised manner might improve the performance of classification solutions. Since for some diseases and symptoms the data used for classification and study is scarce, Generative Adversial Networks can be used to generate more datasets. Transfer learning techniques can also be considered when studying different diseases with similar symptomatology. For instance, if a large data set for RBD or PLM recorded from PD patients is available, an algorithm can be developed and transfered to ALS or MSA sleep studies. The same could be applied for gait or speech abnormality detection.

10.2. Directions for development

Possible Research Directions. The authors have identified some areas that might be worth investigating: (i) Sleep in AD - can show the effect of medication on the lifestyle of the patient; (ii) Differential diagnosis based on speech - speech analysis for motor diseases such as PD, HD, MSA and ALS, lexical analysis for AD and other dementias. Using speech for differential diagnosis can be advantageous as microphones are available in many consumer devices; (iii) Use of EEG - biomarkers can be developed for all neurodegenerative diseases. As EEG technology is non-invasive and less expensive than brain imaging, it can bring advantages and simplify the diagnostic process. Wearable EEG headsets can open even more possibilities in the diagnosis of neurodegenerative diseases; (iv) Memory testing applications - differentiate between AD or other dementing diseases and MCI. Memory tests are currently delivered by medical professionals. By developing applications that focus on the ease of use, the diagnostic process can be simplified and made more accessible, while at the same time allowing for disease tracking; (v) Dual tasking - early onset detection and tracking of dementing and mixed neurodegenerative disorders. By developing tasks that monitor both the cognitive ability and the motor functions of a patient, the progress of disease and risk for further injury can be determined. Dual tasks can be designed with the purpose of measuring the cognitive reserve of a patient. The concept of cognitive reserve is related to the ability of the brain to re-purpose its networks to counter the effects of neurodegeneration. In recent research, the presence of a higher cognitive reserve is indicative of a delayed disease onset or milder symptoms [198].

Applications. Few real-life available applications have been identified by the authors. Usability and adoption by the users was not detailed. Most of the identified applications were developed for PD. These make use of wearable inertial sensors or smartphones. The development is generally limited to one particular disease. As the classification of neurodegenerative diseases is discrete, based on clinical symptoms, the initial diagnosis is difficult as symptoms overlap. As the cause of the disease is not yet known and they cannot yet be treated, disease management focuses on managing the symptoms. An integrative approach looking as neurodegeneration as a continuum could take information from multiple sources (gait, voice, sleep, EEG, brain imaging etc.). This would provide a global view on the disease. Thus allowing a better analysis of the symptoms and a subsequent better treatment management. It might also improve the initial diagnosis. The prediction of disease appearance and onset can also be improved by further developing techniques such as EEG biomarker extraction or sleep characterization. Although brain imaging is a powerful tool in disease diagnosis and monitoring, it is expensive, not easily accessible and might be difficult to use once the disease has advanced significantly. By providing more ubiquitous technologies for tracking, such as wearables, the progression and response to medication might be better observed.

11. Conflict of interest statement.

The authors declare they have no conflict of interest.

References

- M. Prince, A. Wimo, M. Guerchet;, G.-C. Ali, Y.-T. Wu, M. Prina, World Alzheimer Report 2015 - The GLobal Impact of Dementia - An Analysis of Prevalence, Incidence, Cost and Trends, Technical Report, London, 2015.
- [2] S. Przedborski, M. Vila, V. Jackson-Lewis, Neurodegeneration : What is it and where are we ?, J. Clin. Invest. 111 (2003) 3–10. doi:10.1172/JCI200317522.
- [3] C. Laske, H. R. Sohrabi, S. M. Frost, K. López-De-Ipiña, P. Garrard, M. Buscema, J. Dauwels, S. R. Soekadar, S. Mueller, C. Linnemann, S. A. Bridenbaugh, Y. Kanagasingam, R. N. Martins, S. E. O'bryant, Innovative diagnostic tools for early detection of Alzheimer's disease, Alzheimer's and Dementia 11 (2015) 561–578. doi:10.1016/j.jalz.2014.06.004.
- [4] R. Cassani, M. Estarellas, R. San-Martin, F. J. Fraga, T. H. Falk, Systematic review on resting-state EEG for Alzheimer's disease diagnosis and progression assessment, Disease Markers 2018 (2018). doi:10.1155/2018/5174815.
- [5] S. Bhat, U. Rajendra Acharya, N. Dadmehr, H. Adeli, Clinical neurophysiological and automated EEG-based diagnosis of the Alzheimer's disease, European Neurology 74 (2015) 202-210. doi:10.1159/000441447.
- [6] F. Maestú, P. Cuesta, O. Hasan, A. Fernandéz, M. Funke, P. E. Schulz, Corrigendum: The Importance of the Validation of M/EEG With Current Biomarkers in Alzheimer's Disease, Frontiers in Human Neuroscience 13 (2019) 1–10. doi:10.3389/fnhum.2019.00081.

- [7] R. Zhang, G. Simon, F. Yu, Advancing Alzheimer's research: A review of big data promises, International Journal of Medical Informatics 106 (2017) 48-56. URL: http://dx.doi.org/10.1016/j.ijmedinf.2017.07.002. doi:10.1016/j.ijmedinf.2017.07.002.
- [8] A. Alberdi, A. Aztiria, A. Basarab, On the early diagnosis of Alzheimer's Disease from multimodal signals: A survey, Artificial Intelligence in Medicine 71 (2016) 1-29. URL: http://dx.doi.org/10.1016/j.artmed.2016.06.003. doi:10.1016/j.artmed.2016.06.003.
- [9] E. Pellegrini, L. Ballerini, M. d. C. Hernandez, F. M. Chappell, V. González-Castro, D. Anblagan, S. Danso, S. Muñoz-Maniega, D. Job, C. Pernet, G. Mair, T. J. MacGillivray, E. Trucco, J. M. Wardlaw, Machine learning of neuroimaging for assisted diagnosis of cognitive impairment and dementia: A systematic review, Alzheimer's and Dementia: Diagnosis, Assessment and Disease Monitoring 10 (2018) 519–535. doi:10.1016/j.dadm.2018.07.004.
- [10] C. Davatzikos, Machine learning in neuroimaging: Progress and challenges, NeuroImage 197 (2019) 652–656. doi:10.1016/j.neuroimage.2018.10.003.
- [11] N. L. W. Keijsers, M. W. I. M. Horstink, S. C. A. M. Gielen, Online Monitoring of Dyskinesia in Patients with Parkinson's Disease, IEEE Engineering in Medicine and Biology Magazine 22 (2003) 96–103. doi:10.1109/MEMB.2003.1213632.
- [12] S. M. Van Rooden, W. J. Heiser, J. N. Kok, D. Verbaan, J. J. Van Hilten, J. Marinus, The identification of Parkinson's disease subtypes using cluster analysis: A systematic review, Movement Disorders 25 (2010) 969–978. doi:10.1002/mds.23116.
- [13] C. Ahlrichs, M. Lawo, Parkinson's Disease Motor Symptoms in Machine Learning: A Review, Health Informatics - An International Journal 2 (2013) 1-18. URL: http://www.airccse.org/journal/hiij/papers/2413hiij01.pdf. doi:10.5121/hiij.2013.2401. arXiv:1312.3825.
- [14] D. Steins, H. Dawes, P. Esser, J. Collett, Wearable accelerometry-based technology capable of assessing functional activities in neurological populations in community settings: A systematic review, Journal of NeuroEngineering and Rehabilitation 11 (2014) 1–13. doi:10.1186/1743-0003-11-36.
- [15] C. F. Pasluosta, H. Gassner, J. Winkler, J. Klucken, B. M. Eskofier, An Emerging Era in the Management of Parkinson's Disease: Wearable Technologies and the Internet of Things, IEEE Journal of Biomedical and Health Informatics 19 (2015) 1873–1881. doi:10.1109/JBHI.2015.2461555.
- [16] K. J. Kubota, J. A. Chen, M. A. Little, Machine learning for large-scale wearable sensor data in Parkinson's disease: Concepts, promises, pitfalls, and futures, Movement Disorders 31 (2016) 1314–1326. doi:10.1002/mds.26693. arXiv:arXiv:1011.1669v3.
- [17] N. Cummins, A. Baird, B. W. Schuller, Speech analysis for health: Current state-of-the-art and the increasing impact of deep learning, Methods (2018). URL: https://doi.org/10.1016/j.ymeth.2018.07.007. doi:10.1016/j.ymeth.2018.07.007.
- [18] D. Impedovo, G. Pirlo, Dynamic handwriting analysis for the assessment of neurodegenerative diseases: a pattern recognition perspective, IEEE Reviews in Biomedical Engineering 3333 (2018). doi:10.1109/RBME.2018.2840679.
- [19] G. Orrù, W. Pettersson-yeo, A. F. Marquand, G. Sartori, A. Mechelli, Neuroscience and Biobehavioral Reviews Using Support Vector Machine to identify imaging biomarkers of neurological and psychiatric disease : A critical review, Neuroscience and Biobehavioral Reviews 36 (2012) 1140-1152. URL: http://dx.doi.org/10.1016/j.neubiorev.2012.01.004. doi:10.1016/j.neubiorev.2012.01.004.
- [20] J. Figueiredo, C. P. Sant, J. C. Mor, Automatic recognition of gait patterns in human motor disorders using machine learning : A review, Medical Engineering and Physics 53 (2018) 1-12. URL: https://doi.org/10.1016/j.medengphy.2017.12.006. doi:10.1016/j.medengphy.2017.12.006.
- [21] Y. Moon, J. Sung, R. An, M. E. Hernandez, J. J. Sosnoff, Human Movement Science Gait variability in people with neurological disorders : A systematic review and meta-analysis, HUMAN MOVEMENT SCIENCE 47 (2016) 197-208. URL: http://dx.doi.org/10.1016/j.humov.2016.03.010. doi:10.1016/j.humov.2016.03.010.

- [22] B. Tomik, R. J. Guiloff, Dysarthria in amyotrophic lateral sclerosis: A review, Amyotrophic Lateral Sclerosis 11 (2010) 4–15. doi:10.3109/17482960802379004.
- [23] D. T. Plante, Leg actigraphy to quantify periodic limb movements of sleep: A systematic review and meta-analysis, Sleep Medicine Reviews 18 (2014) 425– 434. URL: http://dx.doi.org/10.1016/j.smrv.2014.02.004. doi:10.1016/j.smrv.2014.02.004. arXiv:NIHMS150003.
- [24] G. G. Kovacs, Concepts and classification of neurodegenerative diseases, volume 145, 1 ed., Elsevier B.V., 2017. URL: http://dx.doi.org/10.1016/B978-0-12-802395-2.00021-3. doi:10.1016/B978-0-12-802395-2.00021-3.
- [25] National Institute of Health (NIH), National of Neurological Disorders and Stroke, 2016. URL: http://www.ninds.nih.gov/disorders/epilepsy/detail_epilepsy.htm.
- [26] S. Joshi, P. D. Shenoy, V. Simha, Venugopal, Patnaik, Classification of Neurodegenerative Disorders Based on Major Risk Factors Employing Machine Learning Techniques, IACSIT International Journal of Engineering and Technology 2 (2010).
- [27] B. Dubois, H. H. Feldman, C. Jacova, H. Hampel, J. L. Molinuevo, K. Blennow, S. T. Dekosky, S. Gauthier, D. Selkoe, R. Bateman, S. Cappa, S. Crutch, S. Engelborghs, G. B. Frisoni, N. C. Fox, D. Galasko, M. O. Habert, G. A. Jicha, A. Nordberg, F. Pasquier, G. Rabinovici, P. Robert, C. Rowe, S. Salloway, M. Sarazin, S. Epelbaum, L. C. de Souza, B. Vellas, P. J. Visser, L. Schneider, Y. Stern, P. Scheltens, J. L. Cummings, Advancing research diagnostic criteria for Alzheimer's disease: The IWG-2 criteria, The Lancet Neurology 13 (2014) 614– 629. doi:10.1016/S1474-4422(14)70090-0.
- [28] C. G. Goetz, S. Fahn, P. Martinez-Martin, W. Poewe, C. Sampaio, G. T. Stebbins, M. B. Stern, B. C. Tilley, R. Dodel, B. Dubois, R. Holloway, J. Jankovic, J. Kulisevsky, A. E. Lang, A. Lees, S. Leurgans, P. A. LeWitt, D. Nyenhuis, C. W. Olanow, O. Rascol, A. Schrag, J. A. Teresi, J. J. Van Hilten, N. LaPelle, Movement disorder society-sponsored revision of the unified Parkinson's disease rating scale (MDS-UPDRS): Process, format, and clinimetric testing plan, Movement Disorders 22 (2007) 41–47. doi:10.1002/mds.21198.
- [29] M. M. Hoehn, M. D. Yahr, Parkinsonism : onset, progression, and mortality Parkinsonism: onset, progression, and mortality, Neurology 17 (1967) 427-442. URL: http://www.neurology.org/content/17/5/427.citation. doi:10.1212/WNL.17.5.427.
- [30] G. G. Kovacs, Neuropathology of neurodegenerative diseases: A practical guide, Neuropathology of Neurodegenerative Diseases: A Practical Guide (2014) 1–309. doi:10.1017/CBO9781107588660.
- [31] R. A. Armstrong, On the 'classification' of neurodegenerative disorders: discrete entities, overlap or continuum?, Folia Neuropathologica 3 (2012) 201-218. URL: http://www.termedia.pl/doi/10.5114/fn.2012.30521. doi:10.5114/fn.2012.30521.
- [32] C. U. Onyike, J. Diehl-Schmid, The Epidemiology of Frontotemporal Dementia, Int Rev Psychiatry 25 (2013) 130–137. doi:10.1038/nature13314.A. arXiv:NIHMS150003.
- [33] D. B. Hogan, K. M. Fiest, J. I. Roberts, C. J. Maxwell, J. Dykeman, T. Pringsheim, T. Steeves, E. E. Smith, D. Pearson, N. Jetté, The prevalence and incidence of dementia with lewy bodies: A systematic review, Canadian Journal of Neurological Sciences 43 (2016) S83–S95. doi:10.1017/cjn.2016.2.
- [34] O. B. Tysnes, A. Storstein, Epidemiology of Parkinson's disease, Journal of Neural Transmission 124 (2017) 901–905. doi:10.1007/s00702-017-1686-y.
- [35] A. Katako, P. Shelton, A. L. Goertzen, D. Levin, B. Bybel, M. Aljuaid, H. J. Yoon, D. Y. Kang, S. M. Kim, C. S. Lee, J. H. Ko, Machine learning identified an Alzheimer's disease-related FDG-PET pattern which is also expressed in Lewy body dementia and Parkinson's disease dementia, Scientific Reports 8 (2018) 1–13. URL: http://dx.doi.org/10.1038/s41598-018-31653-6. doi:10.1038/s41598-018-31653-6.
- [36] J. Islam, Y. Zhang, A novel deep learning based multi-class classification method for Alzheimer's disease detection using brain MRI data, Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics) 10654 LNAI (2017) 213-222. doi:10.1007/978-3-319-70772-3-20.

- [37] S. Sarraf, G. Tofighi, Classification of Alzheimer's Disease Structural MRI Data by Deep Learning Convolutional Neural Networks (2016) 8-12. arXiv:1607.06583.
- [38] R. Mahmood, B. Ghimire, Automatic detection and classification of Alzheimer's disease from MRI scans using principal component analysis and artificial neural networks, International Conference on Systems, Signals, and Image Processing (2013) 133–137. doi:10.1109/IWSSIP.2013.6623471.
- [39] C. Davatzikos, S. M. Resnick, X. Wu, P. Parmpi, C. M. Clark, Individual patient diagnosis of AD and FTD via high-dimensional pattern classification of MRI, NeuroImage 41 (2008) 1220–1227. doi:10.1016/j.neuroimage.2008.03.050.
- [40] S. Klöppel, C. M. Stonnington, C. Chu, B. Draganski, R. I. Scahill, J. D. Rohrer, N. C. Fox, C. R. Jack, J. Ashburner, R. S. Frackowiak, Automatic classification of MR scans in Alzheimer's disease, Brain 131 (2008) 681–689. doi:10.1093/brain/awm319.
- [41] A. Kautzky, R. Seiger, A. Hahn, P. Fischer, W. Krampla, S. Kasper, G. G. Kovacs, R. Lanzenberger, Prediction of Autopsy Verified Neuropathological Change of Alzheimer's Disease Using Machine Learning and MRI, Frontiers in Aging Neuroscience 10 (2018) 1–11. doi:10.3389/fnagi.2018.00406.
- [42] C. Plant, S. J. Teipel, A. Oswald, C. Böhm, T. Meindl, J. Mourao-Miranda, A. W. Bokde, H. Hampel, M. Ewers, Automated detection of brain atrophy patterns based on MRI for the prediction of Alzheimer's disease, NeuroImage 50 (2010) 162-174. URL: http://dx.doi.org/10.1016/j.neuroimage.2009.11.046. doi:10.1016/j.neuroimage.2009.11.046.
- [43] C. Salvatore, A. Cerasa, P. Battista, M. C. Gilardi, Magnetic resonance imaging biomarkers for the early diagnosis of Alzheimer 's disease : a machine learning approach 9 (2015) 1–13. doi:10.3389/fnins.2015.00307.
- [44] H. S. Lee, A. H. Bachman, D. Yu, J. Lim, B. A. Ardekani, Predicting progression from mild cognitive impairment to Alzheimer 's disease using longitudinal callosal atrophy 2 (2016) 68–74. doi:10.1016/j.dadm.2016.01.003.
- [45] E. Jabason, M. Omair Ahmad, M. N. Swamy, Deep structural and clinical feature learning for semi-supervised multiclass prediction of Alzheimer's disease, Midwest Symposium on Circuits and Systems 2018-Augus (2019) 791–794. doi:10.1109/MWSCAS.2018.8623946.
- [46] O. B. Ahmed, J. Benois-Pineau, M. Allard, G. Catheline, C. B. Amar, Recognition of Alzheimer's disease and Mild Cognitive Impairment with multimodal image-derived biomarkers and Multiple Kernel Learning, Neurocomputing 220 (2017) 98-110. URL: http://dx.doi.org/10.1016/j.neucom.2016.08.041. doi:10.1016/j.neucom.2016.08.041.
- [47] M. Liu, D. Zhang, D. Shen, Relationship Induced Multi-Template Learning for Diagnosis of Alzheimer's Disease and Mild Cognitive Impairment, IEEE Transactions on Medical Imaging 35 (2016) 1463–1474. doi:10.1109/TMI.2016.2515021.
- [48] D. Lu, K. Popuri, G. W. Ding, R. Balachandar, M. F. Beg, M. Weiner, et. al, Multimodal and Multiscale Deep Neural Networks for the Early Diagnosis of Alzheimer's Disease using structural MR and FDG-PET images, Scientific Reports 8 (2018) 1–13. doi:10.1038/s41598-018-22871-z. arXiv:1710.04782.
- [49] T. Yan, Y. Wang, Z. Weng, W. Du, T. Liu, D. Chen, X. Li, J. Wu, Y. Han, Early-Stage Identification and Pathological Development of Alzheimer's Disease Using Multimodal MRI, Journal of Alzheimer's Disease 68 (2019) 1013–1027. doi:10.3233/JAD-181049.
- [50] A. V. Lebedev, E. Westman, G. J. P. V. Westen, M. G. Kramberger, A. Lundervold, D. Aarsland, NeuroImage : Clinical Random Forest ensembles for detection and prediction of Alzheimer 3 s disease with a good between-cohort robustness 6 (2014) 115–125. doi:10.1016/j.nicl.2014.08.023.
- [51] R. Cassani, T. H. Falk, F. J. Fraga, M. Cecchi, D. K. Moore, R. Anghinah, Towards automated electroencephalography-based Alzheimer's disease diagnosis using portable lowdensity devices, Biomedical Signal Processing and Control 33 (2017) 261-271. URL: http://dx.doi.org/10.1016/j.bspc.2016.12.009. doi:10.1016/j.bspc.2016.12.009.
- [52] L. R. Trambaiolli, N. Spolaôr, A. C. Lorena, R. Anghinah, J. R. Sato, Feature selection before EEG classification supports the diagnosis of Alzheimer's disease, Clinical Neurophysiology 128 (2017) 2058-2067. URL: http://dx.doi.org/10.1016/j.clinph.2017.06.251. doi:10.1016/j.clinph.2017.06.251.

- [53] G. Q. Mamani, F. J. Fraga, G. Tavares, E. Johns, N. D. Phillips, EEG-based biomarkers on working memory tasks for early diagnosis of Alzheimer's disease and mild cognitive impairment, 2017 IEEE Healthcare Innovations and Point of Care Technologies, HI-POCT 2017 2017-Decem (2017) 237-240. doi:10.1109/HIC.2017.8227628.
- [54] M. Dauwan, J. J. van der Zande, E. van Dellen, I. E. Sommer, P. Scheltens, A. W. Lemstra, C. J. Stam, Random forest to differentiate dementia with Lewy bodies from Alzheimer's disease, Alzheimer's and Dementia: Diagnosis, Assessment and Disease Monitoring 4 (2016) 99– 106. URL: http://dx.doi.org/10.1016/j.dadm.2016.07.003. doi:10.1016/j.dadm.2016.07.003.
- [55] D. H. Jeong, Y. D. Kim, I. U. Song, Y. A. Chung, J. Jeong, Wavelet energy and wavelet coherence as eeg biomarkers for the diagnosis of Parkinson's disease-related dementia and Alzheimer's disease, Entropy 18 (2016) 1–17. doi:10.3390/e18010008.
- [56] J. C. McBride, X. Zhao, N. B. Munro, G. A. Jicha, F. A. Schmitt, R. J. Kryscio, C. D. Smith, Y. Jiang, Sugihara causality analysis of scalp EEG for detection of early Alzheimer's disease, NeuroImage: Clinical 7 (2015) 258–265. doi:10.1016/j.nicl.2014.12.005.
- [57] A. König, N. Linz, J. Tröger, M. Wolters, J. Alexandersson, P. Robert, Fully automatic speech-based analysis of the semantic verbal fluency task, Dementia and Geriatric Cognitive Disorders 45 (2018) 198–209. doi:10.1159/000487852.
- [58] K. C. Fraser, J. A. Meltzer, F. Rudzicz, Linguistic features identify Alzheimer's disease in narrative speech, Journal of Alzheimer's Disease 49 (2015) 407–422. doi:10.3233/JAD-150520.
- [59] J. Weiner, M. Engelbart, T. Schultz, Manual and automatic transcriptions in dementia detection from speech, Proceedings of the Annual Conference of the International Speech Communication Association, INTERSPEECH 2017-Augus (2017) 3117–3121. doi:10.21437/Interspeech.2017-112.
- [60] A. König, A. Satt, A. Sorin, R. Hoory, O. Toledo-Ronen, A. Derreumaux, V. Manera, F. Verhey, P. Aalten, P. H. Robert, R. David, Automatic speech analysis for the assessment of patients with predementia and Alzheimer's disease, Alzheimer's and Dementia: Diagnosis, Assessment and Disease Monitoring 1 (2015) 112–124. doi:10.1016/j.dadm.2014.11.012.
- [61] K. Lopez-De-Ipina, U. Martinez-De-Lizarduy, P. M. Calvo, B. Beitia, J. Garcia-Melero, M. Ecay-Torres, A. Estanga, M. Faundez-Zanuy, Analysis of Disfluencies for automatic detection of Mild Cognitive Impartment: A deep learning approach, 2017 International Work Conference on Bio-Inspired Intelligence: Intelligent Systems for Biodiversity Conservation, IWOBI 2017 - Proceedings (2017) 1–4. doi:10.1109/IWOBI.2017.7985526.
- [62] K. López-de Ipiña, J. B. Alonso, C. M. Travieso, J. Solé-Casals, H. Egiraun, M. Faundez-Zanuy, A. Ezeiza, N. Barroso, M. Ecay-Torres, P. Martinez-Lage, U. M. De Lizardui, On the selection of non-invasive methods based on speech analysis oriented to automatic Alzheimer disease diagnosis, Sensors (Switzerland) 13 (2013) 6730–6745. doi:10.3390/s130506730.
- [63] A. Satt, R. Hoory, A. König, P. Aalten, P. H. Robert, Speech-based automatic and robust detection of very early dementia, Proceedings of the Annual Conference of the International Speech Communication Association, INTERSPEECH (2014) 2538–2542.
- [64] I. M. Pavisic, N. C. Firth, S. Parsons, D. M. Rego, T. J. Shakespeare, K. X. Yong, C. F. Slattery, R. W. Paterson, A. J. Foulkes, K. Macpherson, A. M. Carton, D. C. Alexander, J. Shawe-Taylor, N. C. Fox, J. M. Schott, S. J. Crutch, S. Primativo, Eyetracking metrics in young onset Alzheimer's disease: A window into cognitive visual functions, Frontiers in Neurology 8 (2017) 1–16. doi:10.3389/fneur.2017.00377.
- [65] S. Parsons, D. M. Rego, J. Shawe-Taylor, N. C. Firth, S. Primativo, S. J. Crutch, T. J. Shake-speare, C. F. Slattery, K. Macpherson, A. M. Carton, J. M. Schott, D. C. Alexander, R. W. Paterson, A. J. Foulkes, N. C. Fox, I. M. Pavisic, Modelling eye-tracking data to discriminate between Alzheimer's patients and healthy controls, Alzheimer's & Dementia 13 (2017) P597–P598. URL: http://dx.doi.org/10.1016/j.jalz.2017.07.237. doi:10.1016/j.jalz.2017.07.237.
- [66] J. Chung, S. A. Chau, N. Herrmann, K. L. Lanctôt, M. Eizenman, Detection of Apathy in Alzheimer Patients by Analysing Visual Scanning Behaviour with RNNs (2018) 149–157.
- [67] C. Park, J. R. Kim, J. Kim, S. Park, Machine learning-based identification of genetic interactions from heterogeneous gene expression profiles, PLoS ONE 13 (2018) 1–15. doi:10.1371/journal.pone.0201056.

- [68] X. Huang, H. Liu, X. Li, L. Guan, J. Li, L. C. A. M. Tellier, H. Yang, J. Wang, J. Zhang, Revealing Alzheimer's disease genes spectrum in the whole-genome by machine learning, BMC Neurology 18 (2018) 1–8. doi:10.1186/s12883-017-1010-3.
- [69] L. Xu, G. Liang, C. Liao, G. D. Chen, C. C. Chang, An efficient classifier for Alzheimer's disease genes identification, Molecules 23 (2018). doi:10.3390/molecules23123140.
- [70] F. Alvarez, M. Popa, N. Vretos, A. Belmonte-Hernandez, S. Asteriadis, V. Solachidis, T. Mariscal, D. Dotti, P. Daras, Multimodal monitoring of Parkinson's and Alzheimer's patients using the ICT4LIFE platform, 2017 14th IEEE International Conference on Advanced Video and Signal Based Surveillance, AVSS 2017 (2017). doi:10.1109/AVSS.2017.8078527.
- [71] S. J. Colloby, R. A. Cromarty, L. R. Peraza, K. Johnsen, G. Jóhannesson, L. Bonanni, M. Onofrj, R. Barber, J. T. O'Brien, J. P. Taylor, Multimodal EEG-MRI in the differential diagnosis of Alzheimer's disease and dementia with Lewy bodies, Journal of Psychiatric Research 78 (2016) 48–55. doi:10.1016/j.jpsychires.2016.03.010.
- [72] K. C. Fraser, K. Lundholm Fors, D. Kokkinakis, A. Nordlund, An analysis of eyemovements during reading for the detection of mild cognitive impairment (2018) 1016–1026. doi:10.18653/v1/d17-1107.
- [73] M. Grassi, G. Perna, D. Caldirola, K. Schruers, R. Duara, D. A. Loewenstein, A clinicallytranslatable machine learning algorithm for the prediction of Alzheimer's disease conversion in individuals with mild and premild cognitive impairment, Journal of Alzheimer's Disease 61 (2018) 1555–1573. doi:10.3233/JAD-170547.
- [74] M. N. Alam, A. Garg, T. T. K. Munia, R. Fazel-Rezai, K. Tavakolian, Vertical ground reaction force marker for Parkinson's disease, PLoS ONE 12 (2017) 1–13. doi:10.1371/journal.pone.0175951.
- [75] D. G. Zwartjes, T. Heida, J. P. Van Vugt, J. A. Geelen, P. H. Veltink, Ambulatory monitoring of activities and motor symptoms in Parkinsons disease, IEEE Transactions on Biomedical Engineering 57 (2010) 2778–2786. doi:10.1109/TBME.2010.2049573.
- [76] T. Aşuroğlu, K. Açıcı, Ç. Berke Erdaş, M. Kılınç Toprak, H. Erdem, H. Oğul, Parkinson's disease monitoring from gait analysis via foot-worn sensors, Biocybernetics and Biomedical Engineering 38 (2018) 760-772. doi:10.1016/j.bbe.2018.06.002.
- [77] S. Patel, K. Lorincz, R. Hughes, N. Huggins, J. Growdon, D. Standaert, M. Akay, J. Dy, M. Welsh, P. Bonato, Monitoring motor fluctuations in patients with parkinsons disease using wearable sensors, IEEE Transactions on Information Technology in Biomedicine 13 (2009) 864-873. doi:10.1109/TITB.2009.2033471. arXiv:15334406.
- [78] À. Bayés, A. Samá, A. Prats, C. Pérez-López, M. Crespo-Maraver, J. M. Moreno, S. Alcaine, A. Rodriguez-Molinero, B. Mestre, P. Quispe, A. C. de Barros, R. Castro, A. Costa, R. Annicchiarico, P. Browne, T. Counihan, H. Lewy, G. Vainstein, L. R. Quinlan, D. Sweeney, G. ÓLaighin, J. Rovira, D. Rodrigue z Martin, J. Cabestany, A "HOLTER" for Parkinson's disease: Validation of the ability to detect on-off states using the REMPARK system, Gait and Posture 59 (2018) 1–6. URL: http://dx.doi.org/10.1016/j.gaitpost.2017.09.031. doi:10.1016/j.gaitpost.2017.09.031.
- [79] K. Hori, Y. Hirobe, S. Orimo, H. Sawada, A. Inaba, Y. Miyake, Early detection of Parkinson's disease based on gait trajectory analysis using wearable sensors, Journal of the Neurological Sciences 381 (2017) 348. URL: http://linkinghub.elsevier.com/retrieve/pii/S0022510X17314867. doi:10.1016/j.jns.2017.08.989.
- [80] E. Stack, V. Agarwal, R. King, M. Burnett, F. Tahavori, B. Janko, W. Harwin, A. Ashburn, D. Kunkel, Identifying balance impairments in people with Parkinson's disease using video and wearable sensors, Gait and Posture 62 (2018) 321-326. URL: https://doi.org/10.1016/j.gaitpost.2018.03.047. doi:10.1016/j.gaitpost.2018.03.047.
- [81] P. Kugler, C. Jaremenko, J. Schlachetzki, J. Winkler, J. Klucken, B. Eskofier, Automatic recognition of Parkinson's disease using surface electromyography during standardized gait tests, Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBS (2013) 5781–5784. doi:10.1109/EMBC.2013.6610865.

- [82] E. Firtion, C. K. Peng, J. M. Hausdorff, S. L. Mitchell, M. E. Cudkowicz, J. Y. Wei, A. R. Y. L. Goldberger, M. Jeffrey, S. L. Mitchell, C. K. Peng, M. E. Cudkowicz, J. Y. Wei, Altered fractal dynamics of gait : reduced stride-interval correlations with aging and Huntington 's disease (2019).
- [83] A. Zhao, L. Qi, J. Li, J. Dong, H. Yu, A hybrid spatio-temporal model for detection and severity rating of Parkinson's disease from gait data, Neurocomputing 315 (2018) 1-8. URL: https://doi.org/10.1016/j.neucom.2018.03.032. doi:10.1016/j.neucom.2018.03.032.
- [84] A. Samà, C. Pérez-López, D. Rodríguez-Martín, A. Català, J. M. Moreno-Aróstegui, J. Cabestany, E. de Mingo, A. Rodríguez-Molinero, Estimating bradykinesia severity in Parkinson's disease by analysing gait through a waist-worn sensor, Computers in Biology and Medicine 84 (2017) 114-123. URL: http://dx.doi.org/10.1016/j.compbiomed.2017.03.020. doi:10.1016/j.compbiomed.2017.03.020.
- [85] O. Martinez-Manzanera, E. Roosma, M. Beudel, R. W. Borgemeester, T. Van Laar, N. M. Maurits, A Method for Automatic and Objective Scoring of Bradykinesia Using Orientation Sensors and Classification Algorithms, IEEE Transactions on Biomedical Engineering 63 (2016) 1016–1024. doi:10.1109/TBME.2015.2480242.
- [86] M. Bächlin, M. Plotnik, D. Roggen, I. Maidan, J. M. Hausdorff, N. Giladi, G. Tröster, Wearable assistant for Parkinsons disease patients with the freezing of gait symptom, IEEE Transactions on Information Technology in Biomedicine 14 (2010) 436-446. doi:10.1109/TITB.2009.2036165. arXiv:arXiv:1011.1669v3.
- [87] J. Camps, A. Samà, M. Martín, D. Rodríguez-Martín, C. Pérez-López, J. M. Moreno Arostegui, J. Cabestany, A. Català, S. Alcaine, B. Mestre, A. Prats, M. C. Crespo-Maraver, T. J. Counihan, P. Browne, L. R. Quinlan, G. Laighin, D. Sweeney, H. Lewy, G. Vainstein, A. Costa, R. Annicchiarico, A. Bayés, A. Rodríguez-Molinero, Deep learning for freezing of gait detection in Parkinson's disease patients in their homes using a waist-worn inertial measurement unit, Knowledge-Based Systems 139 (2018) 119–131. doi:10.1016/j.knosys.2017.10.017.
- [88] D. Rodríguez-Martín, A. Samà, C. Pérez-López, A. Català, J. M. Arostegui, J. Cabestany, À. Bayés, S. Alcaine, B. Mestre, A. Prats, M. C. Crespo, T. J. Counihan, P. Browne, L. R. Quinlan, G. Laighin, D. Sweeney, H. Lewy, J. Azuri, G. Vainstein, R. Annicchiarico, A. Costa, A. Rodríguez-Molinero, Home detection of freezing of gait using Support Vector Machines through a single waist-worn triaxial accelerometer, PLoS ONE 12 (2017) 1–26. doi:10.1371/journal.pone.0171764.
- [89] S. Mazilu, A. Calatroni, E. Gazit, D. Roggen, J. M. Hausdorff, G. Tröster, Feature learning for detection and prediction of freezing of gait in Parkinson's disease, Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics) 7988 LNAI (2013) 144–158. doi:10.1007/978-3-642-39712-7-11.
- [90] J. Kim, R. Gupta, M. V. Segbroeck, D. Bone, M. P. Black, Z. I. Skordilis, Z. Yang, P. G. Georgiou, S. S. Narayanan, Automatic Estimation of Parkinson's Disease Severity from Diverse Speech Tasks, Proceedings of the Annual Conference of the International Speech Communication Association, INTERSPEECH (2015) 914–918.
- [91] S. Aich, P. M. Pradhan, J. Park, N. Sethi, V. S. S. Vathsa, H. C. Kim, A validation study of freezing of gait (FoG) detection and machine-learning-based FoG prediction using estimated gait characteristics with a wearable accelerometer, Sensors (Switzerland) 18 (2018). doi:10.3390/s18103287.
- [92] A. M. Handojoseno, J. M. Shine, T. N. Nguyen, Y. Tran, S. J. Lewis, H. T. Nguyen, The detection of Freezing of Gait in Parkinson's disease patients using EEG signals based on Wavelet decomposition, Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBS (2012) 69–72. doi:10.1109/EMBC.2012.6345873.
- [93] G. Rigas, A. T. Tzallas, M. G. Tsipouras, P. Bougia, E. E. Tripoliti, D. Baga, D. I. Fotiadis, S. G. Tsouli, S. Konitsiotis, Assessment of tremor activity in the parkinsons disease using a set of wearable sensors, IEEE Transactions on Information Technology in Biomedicine 16 (2012) 478–487. doi:10.1109/TITB.2011.2182616.
- [94] N. Kostikis, D. Hristu-Varsakelis, M. Arnaoutoglou, C. Kotsavasiloglou, A smartphone-based tool for assessing parkinsonian hand tremor, IEEE Journal of Biomedical and Health Informatics 19 (2015) 1835–1842. doi:10.1109/JBHI.2015.2471093.

- [95] M. I. Chelaru, C. Duval, M. Jog, Levodopa-induced dyskinesias detection based on the complexity of involuntary movements, Journal of Neuroscience Methods 186 (2010) 81–89. doi:10.1016/j.jneumeth.2009.10.015.
- [96] M. G. Tsipouras, A. T. Tzallas, D. I. Fotiadis, S. Konitsiotis, On automated assessment of Levodopa-induced dyskinesia in Parkinson's disease, Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBS (2011) 2679– 2682. doi:10.1109/IEMBS.2011.6090736.
- [97] A. Salarian, H. Russmann, C. Wider, P. R. Burkhard, F. J. Vingerhoets, K. Aminian, Quantification of tremor and bradykinesia in Parkinson's disease using a novel ambulatory monitoring system, IEEE Transactions on Biomedical Engineering 54 (2007) 313–322. doi:10.1109/TBME.2006.886670.
- [98] B. T. Cole, S. H. Roy, C. J. De Luca, S. H. Nawab, Dynamical learning and tracking of tremor and dyskinesia from wearable sensors, IEEE Transactions on Neural Systems and Rehabilitation Engineering 22 (2014) 982–991. doi:10.1109/TNSRE.2014.2310904.
- [99] P. Angeles, Y. Tai, N. Pavese, S. Wilson, R. Vaidyanathan, Automated assessment of symptom severity changes during deep brain stimulation (DBS) therapy for Parkinson's disease, IEEE International Conference on Rehabilitation Robotics (2017) 1512–1517. doi:10.1109/ICORR.2017.8009462.
- [100] T. Khan, Parkinson's Disease Assessment Using Speech Anomalies: A Review (2008).
- [101] J. Orozco-Arroyave, F. Honig, J. Arias-Londono, J. Vargas-Bonilla, S. Skoda, J. Rusz, E. Noth, Automatic Detection of Parkinson 's Disease in Running Speech Spoken in Three Different Languages (2014) 1–24.
- [102] M. A. Little, P. E. Mcsharry, E. J. Hunter, J. Spielman, L. O. Ramig, Suitability of Dysphonia Measurements for telemonitoring of Parkinson's Disease, Nature 63 (2009) 502. doi:10.1109/TBME.2008.2005954.
- [103] J. C. Vásquez-Correa, J. Serra, J. R. Orozco-Arroyave, J. F. Vargas-Bonilla, E. Nöth, Effect of acoustic conditions on algorithms to detect Parkinson's disease from speech, ICASSP, IEEE International Conference on Acoustics, Speech and Signal Processing - Proceedings (2017) 5065–5069. doi:10.1109/ICASSP.2017.7953121.
- [104] J. Rusz, R. Cmejla, H. Ruzickova, E. Ruzicka, Quantitative acoustic measurements for characterization of speech and voice disorders in early untreated Parkinson's disease, The Journal of the Acoustical Society of America 129 (2011) 350-367. URL: http://asa.scitation.org/doi/10.1121/1.3514381. doi:10.1121/1.3514381.
- [105] T. Tykalova, J. Rusz, J. Klempir, R. Cmejla, E. Ruzicka, Distinct patterns of imprecise consonant articulation among Parkinson's disease, progressive supranuclear palsy and multiple system atrophy, Brain and Language 165 (2017) 1–9. doi:10.1016/j.bandl.2016.11.005.
- [106] J. Mekyska, E. Janousova, P. Gomez-Vilda, Z. Smekal, I. Rektorova, I. Eliasova, M. Kostalova, M. Mrackova, J. B. Alonso-Hernandez, M. Faundez-Zanuy, K. López-de Ipiña, Robust and complex approach of pathological speech signal analysis, Neurocomputing 167 (2015) 94–111. URL: http://dx.doi.org/10.1016/j.neucom.2015.02.085. doi:10.1016/j.neucom.2015.02.085.
- [107] J. Hlavnika, R. Cmejla, T. Tykalová, K. Šonka, E. Ruzicka, J. Rusz, Automated analysis of connected speech reveals early biomarkers of Parkinson's disease in patients with rapid eye movement sleep behaviour disorder, Scientific Reports 7 (2017) 1–13. doi:10.1038/s41598-017-00047-5.
- [108] B. T. Harel, M. S. Cannizzaro, H. Cohen, N. Reilly, P. J. Snyder, Acoustic characteristics of Parkinsonian speech: A potential biomarker of early disease progression and treatment, Journal of Neurolinguistics 17 (2004) 439–453. doi:10.1016/j.jneuroling.2004.06.001.
- [109] S. Skodda, W. Visser, U. Schlegel, Vowel articulation in Parkinson's disease, Journal of voice: official journal of the Voice Foundation 25 (2011) 467-472. URL: http://www.ncbi.nlm.nih.gov/pubmed/20434876. doi:10.1016/j.jvoice.2010.01.009.
- [110] J. R. Orozco, J. D. Arias, J. F. Vargas, M. C. González Rátiva, E. Nöth, New Spanish speech corpus database for the analysis of people suffering from Parkinson's disease, LREC 2014. Proceedings of the 9th International Conference on Language Resources and Evaluation (2014) 342-347. URL: http://www.lrec-conf.org/proceedings/lrec2014/summaries/7.html. doi:10.1016/S0022391304002549.

- [111] J. Rusz, J. Hlavnicka, T. Tykalova, M. Novotny, P. Dusek, K. Sonka, E. Ruzicka, Smartphone Allows Capture of Speech Abnormalities Associated with High Risk of Developing Parkinson's Disease, IEEE Transactions on Neural Systems and Rehabilitation Engineering 26 (2018) 1495–1507. doi:10.1109/TNSRE.2018.2851787.
- [112] Athanasios Tsanas, Max A. Little, Patrick E. McSharry, Lorraine O. Ramig, Using the cellular mobile telephone network to remotely monitor Parkinsons disease symptom severity, IEEE Transactions on Biomedical Engineering (2012). URL: http://docplayer.net/storage/27/11633800/1504801061/jSpB96EQJYxJnzYVbQHiAg/11633800.pdf.
- [113] R. Das, A comparison of multiple classification methods for diagnosis of Parkinson disease, Expert Systems with Applications 37 (2010) 1568-1572. URL: http://dx.doi.org/10.1016/j.eswa.2009.06.040. doi:10.1016/j.eswa.2009.06.040.
- [114] N. Zhi, B. K. Jaeger, A. Gouldstone, R. Sipahi, S. Frank, Toward Monitoring Parkinson's Through Analysis of Static Handwriting Samples: A Quantitative Analytical Framework, IEEE Journal of Biomedical and Health Informatics 21 (2017) 488–495. doi:10.1109/JBHI.2016.2518858.
- [115] C. Loconsole, G. D. Cascarano, A. Lattarulo, A. Brunetti, G. F. Trotta, D. Buongiorno, I. Bortone, I. D. Feudis, S. I. Tat, G. Losavio, E. D. Sciascio, V. Bevilacqua, A comparison between ANN and SVM classifiers for Parkinson 's disease by using a model-free computerassisted handwriting analysis based on biometric signals, Acepted for pubblication in the Proceedings of the 2018 International Joint Conference on Neural Networks (IJCNN2018) (2018) 1–8. doi:10.1109/IJCNN.2018.8489293.
- [116] C. R. Pereira, D. R. Pereira, G. H. Rosa, V. H. Albuquerque, S. A. Weber, C. Hook, J. P. Papa, Handwritten dynamics assessment through convolutional neural networks: An application to Parkinson's disease identification, Artificial Intelligence in Medicine 87 (2018) 67-77. URL: https://doi.org/10.1016/j.artmed.2018.04.001. doi:10.1016/j.artmed.2018.04.001.
- [117] P. Zham, S. P. Arjunan, S. Raghav, D. K. Kumar, Efficacy of Guided Spiral Drawing in the Classification of Parkinson's Disease, IEEE Journal of Biomedical and Health Informatics 22 (2018) 1648–1652. doi:10.1109/JBHI.2017.2762008.
- [118] P. Drotár, J. Mekyska, I. Rektorová, L. Masarová, Z. Smékal, M. Faundez-Zanuy, Evaluation of handwriting kinematics and pressure for differential diagnosis of Parkinson's disease, Artificial Intelligence in Medicine 67 (2016) 39–46. doi:10.1016/j.artmed.2016.01.004.
- [119] C. Kotsavasiloglou, N. Kostikis, D. Hristu-Varsakelis, M. Arnaoutoglou, Machine learning-based classification of simple drawing movements in Parkinson's disease, Biomedical Signal Processing and Control 31 (2017) 174–180. URL: http://dx.doi.org/10.1016/j.bspc.2016.08.003. doi:10.1016/j.bspc.2016.08.003.
- [120] S. D. Gunnery, E. N. Naumova, M. Saint-Hilaire, L. Tickle-Degnen, Mapping spontaneous facial expression in people with Parkinson's disease: A multiple case study design, Cogent Psychology 4 (2017) 1–15. URL: http://doi.org/10.1080/23311908.2017.1376425. doi:10.1080/23311908.2017.1376425.
- [121] S. R. Livingstone, E. Vezer, L. M. McGarry, A. E. Lang, F. A. Russo, Deficits in the mimicry of facial expressions in Parkinson's disease, Frontiers in Psychology 7 (2016) 1– 12. doi:10.3389/fpsyg.2016.00780.
- [122] A. Bandini, S. Orlandi, H. J. Escalante, F. Giovannelli, M. Cincotta, C. A. Reyes-Garcia, P. Vanni, G. Zaccara, C. Manfredi, Analysis of facial expressions in parkinson's disease through video-based automatic methods, Journal of Neuroscience Methods 281 (2017) 7-20. URL: http://dx.doi.org/10.1016/j.jneumeth.2017.02.006. doi:10.1016/j.jneumeth.2017.02.006.
- [123] A. Cerasa, Machine learning on Parkinson's disease? Let's translate into clinical practice, Journal of Neuroscience Methods 266 (2016) 161-162. URL: http://dx.doi.org/10.1016/j.jneumeth.2015.12.005. doi:10.1016/j.jneumeth.2015.12.005.
- [124] S. Haller, S. Badoud, D. Nguyen, V. Garibotto, K. O. Lovblad, P. R. Burkhard, Individual detection of patients with Parkinson Disease using support vector machine analysis of diffusion tensor imaging data: Initial results, American Journal of Neuroradiology 33 (2012) 2123–2128. doi:10.3174/ajnr.A3126.

- [125] E. Adeli, F. Shi, L. An, C. Y. Wee, G. Wu, T. Wang, D. Shen, Joint feature-sample selection and robust diagnosis of Parkinson's disease from MRI data, NeuroImage 141 (2016) 206–219. doi:10.1016/j.neuroimage.2016.05.054.
- [126] B. Peng, S. Wang, Z. Zhou, Y. Liu, B. Tong, T. Zhang, Y. Dai, A multilevel-ROI-featuresbased machine learning method for detection of morphometric biomarkers in Parkinson's disease, Neuroscience Letters 651 (2017) 88–94. doi:10.1016/j.neulet.2017.04.034.
- [127] T. J. Hirschauer, H. Adeli, J. A. Buford, Computer-Aided Diagnosis of Parkinson's Disease Using Enhanced Probabilistic Neural Network, Journal of Medical Systems 39 (2015). doi:10.1007/s10916-015-0353-9.
- [128] G. Singh, L. Samavedham, Unsupervised learning based feature extraction for differential diagnosis of neurodegenerative diseases: A case study on early-stage diagnosis of Parkinson disease, Journal of Neuroscience Methods 256 (2015) 30-40. URL: http://dx.doi.org/10.1016/j.jneumeth.2015.08.011. doi:10.1016/j.jneumeth.2015.08.011.
- [129] S. Duchesne, Y. Rolland, M. Vérin, Automated Computer Differential Classification in Parkinsonian Syndromes via Pattern Analysis on MRI, Academic Radiology 16 (2009) 61–70. URL: http://dx.doi.org/10.1016/j.acra.2008.05.024. doi:10.1016/j.acra.2008.05.024.
- [130] A. F. Marquand, M. Filippone, J. Ashburner, M. Girolami, J. Mourao-Miranda, G. J. Barker, S. C. Williams, P. N. Leigh, C. R. Blain, Automated, High Accuracy Classification of Parkinsonian Disorders: A Pattern Recognition Approach, PLoS ONE 8 (2013) 1–10. doi:10.1371/journal.pone.0069237.
- [131] C. Salvatore, A. Cerasa, I. Castiglioni, F. Gallivanone, A. Augimeri, M. Lopez, G. Arabia, M. Morelli, M. C. Gilardi, A. Quattrone, Machine learning on brain MRI data for differential diagnosis of Parkinson's disease and Progressive Supranuclear Palsy, Journal of Neuroscience Methods 222 (2014) 230-237. URL: http://dx.doi.org/10.1016/j.jneumeth.2013.11.016. doi:10.1016/j.jneumeth.2013.11.016.
- [132] A. Zhan, M. A. Little, D. A. Harris, S. O. Abiola, E. R. Dorsey, S. Saria, A. Terzis, High Frequency Remote Monitoring of Parkinson's Disease via Smartphone: Platform Overview and Medication Response Detection (2016) 1-12. URL: http://arxiv.org/abs/1601.00960. arXiv:1601.00960.
- [133] E. C. Neto, B. M. Bot, T. Perumal, L. Omberg, J. Guinney, M. Kellen, A. Klein, S. H. Friend, A. D. Trister, Personalized hypothesis tests for detecting medication response in Parkinson Disease patients using iPhone sensor data, Biocomputing 2016 (2016) 273–284. doi:10.1142/97898147494110026.
- [134] W. R. Adams, High-accuracy detection of early Parkinson's Disease using multiple characteristics of finger movement while typing, PLoS ONE 12 (2017) 1–20.
- [135] A. Sadikov, V. Groznik, J. Zabkar, M. Mozina, D. Georgiev, Z. Pirlosek, I. Bralko, Parkinson Check Smart Phone App, Frontiers in Artificial Intelligence and Applications 263 (2014) 1213–1214.
- [136] S. Aghanavesi, D. Nyholm, M. Senek, F. Bergquist, M. Memedi, A smartphonebased system to quantify dexterity in Parkinson's disease patients, Informatics in Medicine Unlocked 9 (2017) 11-17. URL: http://dx.doi.org/10.1016/j.imu.2017.05.005. doi:10.1016/j.imu.2017.05.005.
- [137] L. C. S. Afonso, C. R. Pereira, S. A. T. Weber, C. Hook, J. P. Papa, Parkinson's Disease Identification through Deep Optimum-Path Forest Clustering, Proceedings - 30th Conference on Graphics, Patterns and Images, SIBGRAPI 2017 (2017) 163–169. doi:10.1109/SIBGRAPI.2017.28.
- [138] J. C. Vasquez-Correa, T. Arias-Vergara, J. R. Orozco-Arroyave, B. M. Eskofier, J. Klucken, E. Noth, Multimodal assessment of Parkinson's disease: a deep learning approach, IEEE Journal of Biomedical and Health Informatics 2194 (2018) 1–12. doi:10.1109/JBHI.2018.2866873.
- [139] S. J. Pyo, H. Kim, I. S. Kim, Y.-m. Park, M.-j. Kim, H. M. Lee, S.-b. Koh, Quantitative Gait Analysis in Patients with Huntington 's Disease 10 (2017) 140–144.
- [140] E. Mirek, M. Filip, W. Chwała, K. Banaszkiewicz, Three-Dimensional Trunk and Lower Limbs Characteristics during Gait in Patients with Huntington 's Disease 11 (2017) 1–7. doi:10.3389/fnins.2017.00566.

- [141] A. Mannini, D. Trojaniello, U. D. Croce, A. M. Sabatini, Hidden Markov Model-Based Strategy for Gait Segmentation using Inertial Sensors : Application to Elderly, Hemiparetic Patients and Huntington 's Disease Patients (2015) 5179–5182.
- [142] R. K. Mann, R. Edwards, J. Zhou, A. Fenney, M. Jog, C. Duval, Comparing movement patterns associated with Huntington's chorea and Parkinson's dyskinesia, Experimental Brain Research 218 (2012) 639–654.
- [143] K. Dinesh, M. Xiong, J. Adams, R. Dorsey, G. Sharma, Signal Analysis for Detecting Motor Symptoms in Parkinson's and Huntington's disease using multiple Body-Affixed Sensors: A Pilot Study, 2016 IEEE Western New York Image and Signal Processing Workshop (WNY-ISPW) (????) 1–5. doi:10.1109/WNYIPW.2016.7904834.
- [144] J. Rusz, J. Hlavnicka, R. Cmejla, E. Ruzicka, Automatic evaluation of speech rhythm instability and acceleration in dysarthrias associated with basal 3 (2015) 1–11. doi:10.3389/fbioe.2015.00104.
- [145] M. Novotný, J. Pospíšil, R. Čmejla, J. Rusz, AUTOMATIC DETECTION OF VOICE ON-SET TIME IN DYSARTHRIC SPEECH Czech Technical University in Prague, Faculty of Electrical Engineering, Department of Circuit Theory, Technická 2, 160 00 Prague 6, Czech Republic Charles University in Prague, First Facu (2015) 4340–4344.
- [146] A. Rizk-jackson, D. Stoffers, S. Sheldon, J. Kuperman, A. Dale, J. Goldstein, J. Coreybloom, R. A. Poldrack, A. R. Aron, NeuroImage Evaluating imaging biomarkers for neurodegeneration in pre-symptomatic Huntington's disease using machine learning techniques, NeuroImage 56 (2011) 788-796. URL: http://dx.doi.org/10.1016/j.neuroimage.2010.04.273. doi:10.1016/j.neuroimage.2010.04.273.
- [147] E. Eirola, A. Akusok, K.-m. Björk, H. Johnson, A. Lendasse, Predicting Huntington's Disease : Extreme Learning Machine with Missing Values (2016). doi:10.1007/978-3-319-57421-9.
- [148] S. L. Mason, R. E. Daws, E. Soreq, E. B. Johnson, R. I. Scahill, S. J. Tabrizi, R. A. Barker, A. Hampshire, Predicting Clinical Diagnosis in Huntington 's Disease : An Imaging Polymarker (2018). doi:10.1002/ana.25171.
- [149] N. Georgiou-karistianis, M. A. Gray, J. F. D. Domínguez, A. R. Dymowski, I. Bohanna, L. A. Johnston, A. Churchyard, P. Chua, J. C. Stout, G. F. Egan, Neurobiology of Disease Automated differentiation of pre-diagnosis Huntington 's disease from healthy control individuals based on quadratic discriminant analysis of the basal ganglia : The IMAGE-HD study, Neurobiology of Disease 51 (2013) 82–92. URL: http://dx.doi.org/10.1016/j.nbd.2012.10.001. doi:10.1016/j.nbd.2012.10.001.
- [150] M. D. Tommaso, F. D. Carlo, O. Difruscolo, R. Massafra, V. Sciruicchio, Detection of subclinical brain electrical activity changes in Huntington 's disease using artificial neural networks 114 (2003) 1237-1245. doi:10.1016/S1388-2457(03)00074-9.
- [151] O. F. F. Odish, K. Johnsen, P. V. Someren, R. A. C. Roos, J. G. V. Dijk, EEG may serve as a biomarker in Huntington's disease using machine learning automatic classification (2018) 1-8. doi:10.1038/s41598-018-34269-y.
- [152] J. M. Hausdorff, A. Lertratanakul, M. E. Cudkowicz, D. Kaliton, A. L. Peterson, A. L. Goldberger, Dynamic markers of altered gait rhythm in amyotrophic lateral sclerosis, Journal of Applied Physiology 88 (2017) 2045–2053. doi:10.1152/jappl.2000.88.6.2045.
- [153] S. Dutta, A. Chatterjee, S. Munshi, An automated hierarchical gait pattern identification tool employing cross-correlation-based feature extraction and recurrent neural network based classification, Expert Systems 26 (2009) 202–217. doi:10.1111/j.1468-0394.2009.00479.x.
- [154] Y. Xia, Q. Gao, Y. Lu, Q. Ye, A novel approach for analysis of altered gait variability in amyotrophic lateral sclerosis, Medical and Biological Engineering and Computing 54 (2016) 1399–1408. doi:10.1007/s11517-015-1413-5.
- [155] Y. Yunusova, N. L. Graham, S. Shellikeri, K. Phuong, M. Kulkarni, E. Rochon, D. F. Tang-Wai, T. W. Chow, S. E. Black, L. H. Zinman, J. R. Green, Profiling Speech and Pausing in Amyotrophic Lateral Sclerosis (ALS) and Frontotemporal Dementia (FTD), Plos One 11 (2016) e0147573. doi:10.1371/journal.pone.0147573.

- [156] J. Wang, P. V. Kothalkar, B. Cao, D. Heitzman, Towards automatic detection of amyotrophic lateral sclerosis from speech acoustic and articulatory samples, Proceedings of the Annual Conference of the International Speech Communication Association, INTERSPEECH 08-12-Sept (2016) 1195–1199. doi:10.21437/Interspeech.2016-1542.
- [157] A. Bandini, J. R. Green, B. Taati, S. Orlandi, L. Zinman, Y. Yunusova, Automatic detection of amyotrophic lateral sclerosis (ALS) from video-based analysis of facial movements: Speech and non-speech tasks, Proceedings - 13th IEEE International Conference on Automatic Face and Gesture Recognition, FG 2018 (2018) 150–157. doi:10.1109/FG.2018.00031.
- [158] T. Fekete, N. Zach, L. R. Mujica-Parodi, M. R. Turner, Multiple kernel learning captures a systems-level functional connectivity biomarker signature in amyotrophic lateral sclerosis, PLoS ONE 8 (2013) 1–8. doi:10.1371/journal.pone.0085190.
- [159] H. K. van der Burgh, R. Schmidt, H. J. Westeneng, M. A. de Reus, L. H. van den Berg, M. P. van den Heuvel, Deep learning predictions of survival based on MRI in amyotrophic lateral sclerosis, NeuroImage: Clinical 13 (2017) 361-369. URL: http://dx.doi.org/10.1016/j.nicl.2016.10.008. doi:10.1016/j.nicl.2016.10.008.
- [160] X. Zhang, P. E. Barkhaus, W. Z. Rymer, P. Zhou, Machine learning for supporting diagnosis of amyotrophic lateral sclerosis using surface electromyogram, IEEE Transactions on Neural Systems and Rehabilitation Engineering 22 (2014) 96–103. doi:10.1109/TNSRE.2013.2274658.
- [161] Y. Eun, J. Park, M. Kyung, S. Eun, J. Kim, Y. Jeong, H.-t. Kim, J. Whan, Brain & Language Differences in early speech patterns between Parkinson variant of multiple system atrophy and Parkinson 's disease, Brain and Language 147 (2015) 14–20. URL: http://dx.doi.org/10.1016/j.bandl.2015.04.007. doi:10.1016/j.bandl.2015.04.007.
- [162] J. Soli-soler, R. Janc, J. A. Fiz, J. Morera, Towards automatic pitch detection in snoring signals (2000) 2974–2976.
- [163] B. Högl, A. Stefani, A. Videnovic, Idiopathic REM sleep behaviour disorder and neurodegeneration - An update, Nature Reviews Neurology 14 (2018) 40-56. URL: http://dx.doi.org/10.1038/nrneurol.2017.157. doi:10.1038/nrneurol.2017.157.
- [164] E. K. St Louis, B. F. Boeve, REM Sleep Behavior Disorder: Diagnosis, Clinical Implications, and Future Directions, Mayo Clinic Proceedings 92 (2017) 1723-1736. URL: https://doi.org/10.1016/j.mayocp.2017.09.007. doi:10.1016/j.mayocp.2017.09.007.
- [165] R. J. Brazete, J.-f. Gagnon, R. B. Postuma, J.-a. Bertrand, D. Petit, J. Montplaisir, Neurobiology of Aging Electroencephalogram slowing predicts neurodegeneration in rapid eye movement sleep behavior disorder, Neurobiology of Aging (2015) 1-8. URL: http://dx.doi.org/10.1016/j.neurobiolaging.2015.10.007. doi:10.1016/j.neurobiolaging.2015.10.007.
- [166] G. Ruffini, D. Iba, E. Kroupi, J. Montplaisir, R. B. Postuma, M. Castellano, A. Soria-frisch, Algorithmic complexity of EEG for prognosis of neurodegeneration in idiopathic rapid eye movement behavior disorder (RBD) (2018) 1–9.
- [167] L. M. Berrada, A. Deghan, D. G. Marchand, R. Postuma, J. Montplaisir, K. Jeribi, J.-F. Gagnon, Application of machine-learning methods for identification of cognitive markers of dementia in REM sleep behavior disorder, Sleep Medicine 40 (2017) e102. doi:10.1016/j.sleep.2017.11.297.
- [168] R. Ferri, F. Rundo, M. Manconi, G. Plazzi, O. Bruni, A. Oldani, L. Ferini-strambi, M. Zucconi, Improved computation of the atonia index in normal controls and patients with REM sleep behavior disorder, Sleep Medicine 11 (2010) 947–949. URL: http://dx.doi.org/10.1016/j.sleep.2010.06.003. doi:10.1016/j.sleep.2010.06.003.
- [169] M. Cesari, J. A. E. Christensen, L. Kempfner, A. N. Olesen, G. Mayer, K. Kesper, W. H. Oertel, F. Sixel-döring, C. Trenkwalder, H. B. D. Sorensen, P. Jennum, Comparison of computerized methods for rapid eye movement sleep without atonia detection (2018) 1–11. doi:10.1093/sleep/zsy133.
- [170] J. Kempfner, H. B. D. Sorensen, M. Nikolic, P. Jennum, Early Automatic Detection of Parkinson's Disease Based on Sleep Recordings, American Clinical Neurophisiology Society 31 (2014) 409–415.

- [171] G. Ruffini, M. Castellano, D. Ibañez, L. Dubreuil, Deep learning with EEG spectrograms in rapid eye movement behavior disorder (2018).
- [172] B. Högl, A. Stefani, Restless legs syndrome and periodic leg movements in patients with movement disorders: Specific considerations, Movement Disorders 32 (2017) 669–681. doi:10.1002/mds.26929.
- [173] R. Ferri, F. Rundo, A. Silvani, M. Zucconi, D. Aricò, O. Bruni, F. I. Cosentino, L. Ferrin-Strambi, M. Manconi, Sequence analysis of leg movements during sleep with different intervals (<10, 10–90 and >90 s) in restless legs syndrome, Journal of Sleep Research 26 (2017) 436– 443. doi:10.1111/jsr.12500.
- [174] F. Ferrillo, M. Beelke, P. Canovaro, T. Watanabe, D. Aricò, P. Rizzo, S. Garbarino, L. Nobili, F. De Carli, Changes in cerebral and autonomic activity heralding periodic limb movements in sleep, Sleep Medicine 5 (2004) 407–412. doi:10.1016/j.sleep.2004.01.008.
- [175] M. Sieminski, J. Pyrzowski, M. Partinen, Periodic limb movements in sleep are followed by increases in EEG activity, blood pressure, and heart rate during sleep, Sleep and Breathing 21 (2017) 497–503. doi:10.1007/s11325-017-1476-7.
- [176] D. Tkach, H. Huang, T. A. Kuiken, Study of stability of time-domain features for electromyographic pattern recognition, Journal of NeuroEngineering and Rehabilitation 7 (2010) 1–13. doi:10.1186/1743-0003-7-21.
- [177] I. Umut, G. Çentik, Detection of Periodic Leg Movements by Machine Learning Methods Using Polysomnographic Parameters Other Than Leg Electromyography, Computational and Mathematical Methods in Medicine 2016 (2016). doi:10.1155/2016/2041467.
- [178] E. Sforza, M. Johannes, B. Claudio, The PAM-RL ambulatory device for detection of periodic leg movements: A validation study, Sleep Medicine 6 (2005) 407–413. doi:10.1016/j.sleep.2005.01.004.
- [179] T. Shochat, A. Oksenberg, N. Hadas, A. Molotsky, P. Lavie, The KickStripTM: A Novel Testing Device for Periodic Limb Movement Disorder, Sleep 26 (2003) 480–483. doi:10.1093/sleep/26.4.480.
- [180] Y. Athavale, A. Raissi, S. Krishnan, H. Nouraei, M. I. Boulos, D. D. Dopsa, B. J. Murray, A. G. Berneshawi, Advanced signal analysis for the detection of periodic limb movements from bilateral ankle actigraphy, Journal of Sleep Research 26 (2016) 14–20. doi:10.1111/jsr.12438.
- [181] C. R. Jack, M. A. Bernstein, N. C. Fox, P. Thompson, G. Alexander, D. Harvey, et. al, The Alzheimer's Disease Neuroimaging Initiative (ADNI): MRI methods, Journal of Magnetic Resonance Imaging 27 (2008) 685–691. doi:10.1002/jmri.21049.
- [182] E. E. Bron, M. Smits, W. M. van der Flier, H. Vrenken, F. Barkhof, P. Scheltens, et. al, Standardized evaluation of algorithms for computer-aided diagnosis of dementia based on structural MRI: The CADDementia challenge, NeuroImage 111 (2015) 562-579. URL: http://dx.doi.org/10.1016/j.neuroimage.2015.01.048. doi:10.1016/j.neuroimage.2015.01.048.
- [183] F. Boller, J. Becker, Dementia-bank database guide, University of Pittsburgh (2005).
- [184] S. Frenkel-Toledo, N. Giladi, C. Peretz, T. Herman, L. Gruendlinger, J. M. Hausdorff, Effect of gait speed on gait rhytmicity in Parkinson's disease: variability of stride time and swing time respong differently, Journal of neuroengineering and rehabilitation 2 (2005) 15. doi:10.1186/1743-Received. arXiv:NIHMS150003.
- [185] P. LaMontagne, S. Keefe, W. Lauren, C. Xiong, E. Grant, K. Moulder, J. Morris, T. Benzinger, D. Marcus, OASIS-3: Longitudinal neuroimaging, clinical and cognitive dataset for normal aging and alzheimer's disease, The Journal of the Alzheimer's Association 14 (2018) 1097.
- [186] P. Drotár, J. Mekyska, I. Rektorová, L. Masarová, Z. Smékal, M. Faundez-Zanuy, Analysis of in-air movement in handwriting: A novel marker for Parkinson's disease, Computer Methods and Programs in Biomedicine 117 (2014) 405–411. doi:10.1016/j.cmpb.2014.08.007.
- [187] B. E. Sakar, M. E. Isenkul, C. O. Sakar, A. Sertbas, F. Gurgen, S. Delil, H. Apaydin, O. Kursun, Collection and analysis of a Parkinson speech dataset with multiple types of sound recordings, IEEE Journal of Biomedical and Health Informatics 17 (2013) 828-834. doi:10.1109/JBHI.2013.2245674.

- [188] M. E. Isenkul, B. E. Sakar, O. Kursun, Improved spiral test using digitized graphics tablet for monitoring Parkinson's disease (2014) 171–175. doi:10.13140/RG.2.1.1898.6005.
- [189] C. Taleb, M. Khachab, C. Mokbel, L. Likforman-Sulem, Feature selection for an improved Parkinson's disease identification based on handwriting (2017) 52–56. doi:10.1109/asar.2017.8067759.
- [190] K. Marek, D. Jennings, S. Lasch, A. Siderowf, C. Tanner, T. Simuni, et. al, The Parkinson Progression Marker Initiative (PPMI), Progress in Neurobiology 95 (2011) 629–635. doi:10.1016/j.pneurobio.2011.09.005.
- [191] J. S. Paulsen, D. R. Langbehn, J. C. Stout, E. Aylward, C. A. Ross, M. Nance, M. Guttman, S. Johnson, M. MacDonald, L. J. Beglinger, K. Duff, E. Kayson, K. Biglan, I. Shoulson, D. Oakes, M. Hayden, Detection of Huntington's disease decades before diagnosis: The Predict-HD study, Journal of Neurology, Neurosurgery and Psychiatry 79 (2008) 874–880. doi:10.1136/jnnp.2007.128728.
- [192] S. J. Tabrizi, D. R. Langbehn, B. R. Leavitt, R. A. Roos, A. Durr, D. Craufurd, C. Kennard, S. L. Hicks, N. C. Fox, R. I. Scahill, B. Borowsky, A. J. Tobin, H. D. Rosas, H. Johnson, R. Reilmann, B. Landwehrmeyer, J. C. Stout, Biological and clinical manifestations of Huntington's disease in the longitudinal TRACK-HD study: crosssectional analysis of baseline data, The Lancet Neurology 8 (2009) 791-801. URL: http://dx.doi.org/10.1016/S1474-4422(09)70170-X. doi:10.1016/S1474-4422(09)70170-X.
- [193] H. M. Gray, L. Tickle-Degnen, A meta-analysis of performance on emotion recognition tasks in Parkinson's disease., Neuropsychology 24 (2010) 176–191. doi:10.1037/a0018104.
- [194] S. Mazilu, M. Hardegger, Z. Zhu, D. Roggen, G. Troester, M. Plotnik, J. Hausdorff, Online Detection of Freezing of Gait with Smartphones and Machine Learning Techniques, Proceedings of the 6th International Conference on Pervasive Computing Technologies for Healthcare (2012) 123-130. URL: http://eudl.eu/doi/10.4108/icst.pervasivehealth.2012.248680. doi:10.4108/icst.pervasivehealth.2012.248680.
- [195] M. Dyrba, F. Barkhof, A. Fellgiebel, M. Filippi, L. Hausner, K. Hauenstein, T. Kirste, S. J. Teipel, Predicting Prodromal Alzheimer 's Disease in Subjects with Mild Cognitive Impairment Using Machine Learning Classification of Multimodal Multicenter Diffusion-Tensor and Magnetic Resonance Imaging Data (2015) 1–10. doi:10.1111/jon.12214.
- [196] Z. Galaz, J. Mekyska, Z. Mzourek, Z. Smekal, I. Rektorova, I. Eliasova, M. Kostalova, M. Mrackova, D. Berankova, Prosodic analysis of neutral, stress-modified and rhymed speech in patients with Parkinson's disease, Computer Methods and Programs in Biomedicine 127 (2016) 301-317. URL: http://dx.doi.org/10.1016/j.cmpb.2015.12.011. doi:10.1016/j.cmpb.2015.12.011.
- [197] C. Melissant, A. Ypma, E. E. Frietman, C. J. Stam, A method for detection of Alzheimer's disease using ICA-enhanced EEG measurements, Artificial Intelligence in Medicine 33 (2005) 209–222. doi:10.1016/j.artmed.2004.07.003.
- [198] A. Menardi, A. Pascual-Leone, P. Fried, E. Santarnecchi, The role of cognitive reserve in Alzheimer's disease and aging. A multi-modal imaging review, Journal Alzheimer's Disease (2018) 1341–1362. doi:10.3233/JAD-180549.